



CREATES Research Paper 2009-21

Multipower Variation for Brownian Semistationary Processes

Ole E. Barndorff-Nielsen, José Manuel Corcuera and Mark Podolskij

School of Economics and Management Aarhus University Bartholins Allé 10, Building 1322, DK-8000 Aarhus C Denmark

Multipower Variation for Brownian Semistationary Processes^{*}

Ole E. Barndorff-Nielsen University of Aarhus and CREATES[†] José Manuel Corcuera University of Barcelona[‡]

Mark Podolskij ETH Zürich and CREATES §

May 26, 2009

Abstract

In this paper we study the asymptotic behaviour of power and multipower variations of processes Y:

$$Y_t = \int_{-\infty}^t g(t-s)\sigma_s W(\mathrm{d}s) + Z_t$$

where $g: (0, \infty) \to \mathbb{R}$ is deterministic, $\sigma > 0$ is a random process, W is the stochastic Wiener measure, and Z is a stochastic process in the nature of a drift term. Processes of this type serve, in particular, to analyse data of velocity increments of a fluid in a turbulence regime with spot intermittency σ . The purpose of the present paper is to determine the probabilistic limit behaviour of the (multi)power variations of Y, as a basis for studying properties of the intermittency process σ . Notably the processes Y are in general not of the semimartingale kind and the established theory of multipower variation for semimartingales does not suffice for deriving the limit properties. As a key tool for the results a general central limit theorem for triangular Gaussian schemes is formulated and proved. Examples and an application to realised variance ratio are given.

Keywords: Central Limit Theorem; Gaussian Processes; Intermittency; Nonsemimartingales; Turbulence; Volatility; Wiener Chaos.

JEL Classification: C10; C80.

^{*}Ole E. Barndorff-Nielsen and Mark Podolskij acknowledge financial support from CREATES, funded by the Danish National Research Foundation, and from the Thiele Centre. The work of José Manuel Corcuera is supported by the MEC Grant No. MTM2006-03211.

[†]Department of Mathematical Sciences, University of Aarhus, Ny Munkegade, DK–8000 Aarhus C, Denmark. e-mail: oebn@imf.au.dk

[‡]Universitat de Barcelona, Gran Via de les Corts Catalanes 585, 08007 Barcelona, Spain. e-mail: jmcorcuera@ub.edu.

[§]Department of Mathematics, ETH Zürich, HG G32.2, 8092 Zürich, Switzerland, Email: mark.podolskij@math.ethz.ch.

1 Introduction

The motivation for the development of the results reported in this paper has been the need to construct tools for studying the probabilistic limit behaviour of (realised) quadratic variation and other multipower variations in relation to the class of *Brownian semistationary* (\mathcal{BSS}) processes. This class, which was introduced in [BNSch09], consists of the processes $Y = \{Y_t\}_{t \in \mathbb{R}}$ that are defined by

$$Y_t = \mu + \int_{-\infty}^t g(t-s)\sigma_s W(\mathrm{d}s) + \int_{-\infty}^t q(t-s)a_s \mathrm{d}s \tag{1.1}$$

where μ is a constant, W is the stochastic Wiener measure, g and q are nonnegative deterministic functions on \mathbb{R} , with g(t) = q(t) = 0 for $t \leq 0$, and σ and a are càdlàg processes. When σ and a are stationary then so is Y. Hence the name Brownian semistationary processes. The \mathcal{BSS} processes form the natural analogue, for stationarity related processes, to the class \mathcal{BSM} of Brownian semimartingales

$$Y_t = \mu + \int_0^t \sigma_s \mathrm{d}W_s + \int_0^t a_s \mathrm{d}s.$$
(1.2)

In the context of stochastic modelling in finance and in turbulence the process σ embodies the *volatility* or *intermittency* of the dynamics, whether the framework is that of \mathcal{BSM} or \mathcal{BSS} . For detailed discussion of \mathcal{BSS} and the more general concept of tempo-spatial ambit processes see [BNSch04], [BNSch07], [BNSch08a], [BNSch08b], [BNSch08c], [BNSch09]. Such processes are, in particular, able to reproduce key stylized features of turbulent data.

A main difference between \mathcal{BSM} and \mathcal{BSS} is that in general models of the \mathcal{BSS} form are not semimartingales (for a discussion of this, see Section 3 of [BNSch09]). In consequence, various important techniques developed for semimartingales, such as the calculation of quadratic variation by Ito algebra and those of multipower variation, do not apply or suffice in \mathcal{BSS} settings. The present paper addresses some of the issues that this raises.

The theory of multipower variation was primarily developed as a basis for inference on σ under \mathcal{BSM} models and, more generally Ito semimartingales, with particular focus on inference about the integrated squared volatility σ^{2+} given by

$$\sigma_t^{2+} = \int_0^t \sigma_s^2 \mathrm{d}s. \tag{1.3}$$

This quantity is likewise a focal point for the results discussed in the following.

Section 2 introduces common notation for multipower variation and recalls some basic properties of such quantities. A law of large numbers and a central limit theorem for multipower variation of triangular arrays of Gaussian random variables are derived in Section 3, and these limit results are drawn upon in Section 4 to establish probability and central limit theorems for multipower variation for \mathcal{BSS} processes, with most of the proofs postponed to the penultimate Section 7. Section 5 presents several examples, and Section 6 discusses an application concerning the limit behaviour of the realised variation ratio, i.e. the ratio of realised bipower variation to realised quadratic variation. The final Section 8 concludes and indicates some possible directions for further related work. An appendix summarises a number of properties of the modified Bessel functions of the modified third kind, needed for derivations in Section 5.

2 Multipower Variation

The concept of (realised) multipower variation was originally introduced in [BNS04a] in the context of semimartingales, and the mathematical theory has been studied in a number of papers ([BNGJPS06], [Jac08a], [BNSW06], [KiPo08]) while various applications are the main subjects in ([BNS04b]), [BNS06], [BNS07], [Jac08b], [Woe06]. Multipower variation turns out to be useful for analysing properties of parts of a process that are not directly observable. In this section we present the definition of realised multipower variation and recall its asymptotic properties for some classes of processes.

Let us consider a continuous-time process X, defined on some filtered probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t\geq 0}, P)$, that is observed at equidistant time points $t_i = i/n, i = 0, \ldots, [nt]$. A realised multipower variation of the process X is an object of the type

$$\sum_{i=1}^{[nt]-k+1} \prod_{j=1}^{k} |\Delta_{i+j-1}^{n} X|^{p_{j}} , \qquad \Delta_{i}^{n} X = X_{\frac{i}{n}} - X_{\frac{i-1}{n}} , \qquad p_{1}, \dots, p_{k} \ge 0 , \qquad (2.1)$$

for some fixed number $k \ge 1$. We now present an overview of the asymptotic theory for quantities of the form (2.1) for various types of processes X.

We start with the \mathcal{BSM} case

$$X_{t} = X_{0} + \int_{0}^{t} a_{s} ds + \int_{0}^{t} \sigma_{s} dW_{s}$$
(2.2)

where W is a Brownian motion, a is a locally bounded and predictable drift process, and σ is an adapted and càdlàg volatility process. As was established in [BNGJPS06], the convergence in probability

$$n^{p^+/2-1} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n X|^{p_j} \xrightarrow{ucp} \mu_{p_1} \cdots \mu_{p_k} \int_0^t |\sigma_s|^{p+} ds$$
(2.3)

holds, where $p_+ = \sum_{j=1}^k p_j$ and $\mu_p = E[|u|^p]$, $u \sim N(0,1)$ and we write $Z^n \xrightarrow{ucp} Z$ when $\sup_{t \in [0,T]} |Z_t^n - Z_t| \longrightarrow 0$ for any T > 0. Under a further condition on the volatility process one obtains the associated stable central limit theorem:

$$\sqrt{n} \left(n^{p^+/2-1} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n X|^{p_j} - \mu_{p_1} \cdots \mu_{p_k} \int_0^t |\sigma_s|^{p+} ds \right) \xrightarrow{st} \sqrt{A} \int_0^t |\sigma_s|^{p+} dB_s , \quad (2.4)$$

where B is another Brownian motion, defined on an extension of the probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t\geq 0}, P)$ and independent of \mathcal{F} , and the constant A is given by

$$A = \prod_{l=1}^{k} \mu_{2p_l} - (2k-1) \prod_{l=1}^{k} \mu_{p_l}^2 + 2 \sum_{m=1}^{k-1} \prod_{l=1}^{m} \mu_{p_l} \prod_{l=k-m+1}^{k} \mu_{p_l} \prod_{l=1}^{k-m} \mu_{p_l+p_{l+m}}$$

Recall that the stable convergence of processes is defined as follows. A sequence of processes Z^n converges stably in law towards the process Z (written $Z^n \xrightarrow{st} Z$), that is defined on the extension of the original probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t\geq 0}, P)$, if and only if for any bounded and continuous real-valued functional f and any \mathcal{F} -measurable random variable V it holds that

$$\lim_{n \to \infty} E[f(Z^n)V] = E[f(Z)V].$$

When the latter holds only for \mathcal{G} -measurable random variables V, where \mathcal{G} is a sub- σ -algebra of \mathcal{F} , we speak of \mathcal{G} -stable convergence and use the notation $Z^n \xrightarrow{\mathcal{G}-st} Z$.

A crucial property of the realised multipower variation is its robustness to jumps when $\max_i(p_i) < 2$ ([BNSW06], [Jac08c]). Assume for a moment that X is a general Ito semimartingale with continuous part X^c satisfying (2.2). Then, by (2.3) and the robustness property, we obtain the convergence

$$\frac{\mu_1^{-2} \sum_{i=1}^{[nt]-1} |\Delta_i^n X| |\Delta_{i+1}^n X|}{\sum_{i=1}^{[nt]} |\Delta_i^n X|^2} \xrightarrow{P} \frac{[X^c]}{[X]} ,$$

where [X] denotes the quadratic variation of the semimartingale X and the limit is less than or equal to 1. The latter result, together with the stable convergence in (2.4), can be used to construct a formal test for jumps (see [BNS04a]). On the other hand, we know that if the limit of the left-hand side is greater than 1 (which is the case for some typical turbulence data), the process X can not be an Ito semimartingale.

In another direction, a study [BNCPW09] was made of the asymptotic behaviour of multipower variation for processes of the type

$$X_t = X_0 + \int_0^t \sigma_s dG_s , \qquad t \ge 0 ,$$
 (2.5)

where G is a continuous Gaussian process with centered and stationary increments (the latter integral is defined as a Riemann-Stieltjes integral). The process defined in (2.5) is, in general, also not a semimaringale, and the theory in [BNGJPS06] does not apply. In particular, a different normalisation is required. Define the (normalised) multipower variation by

$$V(X, p_1, \dots, p_k)_t^n = \frac{1}{n\tau_n^{p_+}} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n X|^{p_j} , \qquad p_1, \dots, p_k \ge 0 ,$$

where $\tau_n > 0$ is given by

$$\tau_n^2 = \bar{R}(1/n) \tag{2.6}$$

with

$$\bar{R}(t) = E[(G_{s+t} - G_s)^2].$$
(2.7)

Under some assumptions on \overline{R} and the volatility process σ it was shown that

$$V(X, p_1, \dots, p_k)_t^n \xrightarrow{ucp} \rho_{p_1, \dots, p_k} \int_0^t |\sigma_s|^{p+} ds$$

for a certain constant ρ_{p_1,\ldots,p_k} which depends on the behaviour of \overline{R} near 0. Furthermore, an associated (stable) central limit theorem, of a form similar to (2.4), was derived. Note however that in general there are essential differences between the characters of \mathcal{BSS} processes and processes of type (2.5).

3 Multipower variation of Gaussian triangular arrays

In this subsection we derive some asymptotic results for functionals of arrays of stationary Gaussian sequences. We consider a triangular array $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$ (t > 0) of row-wise stationary Gaussian variables with mean 0 and variance 1. Let

$$r_n(j) = \operatorname{cor}(X_{1,n}, X_{1+j,n}) , \qquad j \ge 0,$$
(3.1)

be the correlation function of $(X_{i,n})_{1 \le i \le [nt]}$. Assume that the array $(X_{i,n})_{n \ge 1, 1 \le i \le [nt]}$ is "nondegenerate", i.e. the covariance matrix of $(X_{i,n}, \ldots, X_{i+k,n})$ is invertible for any $k \ge 1$ and $n \ge 1$ (otherwise the results below do not hold).

Now, define the multipower variation associated with the sequence $(X_{i,n})_{n>1,1\leq i\leq [nt]}$:

$$V(p_1,\ldots,p_k)_t^n = \frac{1}{n} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |X_{i+j-1,n}|^{p_j}, \qquad p_1,\ldots,p_k \ge 0.$$
(3.2)

Our first result is the weak law of large numbers.

Theorem 1 Assume that there exists a sequence r(j) with

$$r_n^2(j) \le r(j) , \qquad \frac{1}{n} \sum_{j=1}^{n-1} r(j) \to 0$$
 (3.3)

as $n \to \infty$. Then it holds that

$$V(p_1,\ldots,p_k)_t^n - \rho_{p_1,\ldots,p_k}^{(n)} t \xrightarrow{ucp} 0 , \qquad (3.4)$$

where

$$\rho_{p_1,\dots,p_k}^{(n)} = E\Big[|X_{1,n}|^{p_1}\cdots|X_{k,n}|^{p_k}\Big].$$
(3.5)

Proof of Theorem 1: See Section 7.

Before we present the associated central limit theorem we need to introduce another Gaussian process. Suppose that $r_n(j) \to \rho(j)$, j = 1, ..., k-1, for some numbers $\rho(j)$. Let $(Q_i)_{i\geq 1}$ be a non-degenerate stationary centered (discrete time) Gaussian process with expectation 0, variance 1 and correlation function

$$\rho(j) = \operatorname{cor}(Q_1, Q_{1+j}) , \qquad j \ge 1.$$
(3.6)

Define

$$V_Q(p_1,\ldots,p_k)_t^n = \frac{1}{n} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |Q_{i+j-1}|^{p_j}$$
(3.7)

and let $\rho_{p_1,\dots,p_k} = E(|Q_1|^{p_1}\cdots |Q_k|^{p_k})$. Then $\rho_{p_1,\dots,p_k}^{(n)} \to \rho_{p_1,\dots,p_k}$ and in this case we obtain the ucp convergence

$$V(p_1,\ldots,p_k)_t^n \xrightarrow{ucp} \rho_{p_1,\ldots,\cdot p_k} t.$$

The multivariate central limit theorem for the family $(V(p_1^j, \ldots, p_k^j)_t^n)_{1 \le j \le d}$ is as follows:

Theorem 2 Assume that

$$r_n(j) \to \rho(j) , \qquad j \ge 0,$$

$$(3.8)$$

where $\rho(j)$ is the correlation function of some stationary centered discrete time Gaussian process $(Q_i)_{i\geq 1}$ with $E[Q_i^2] = 1$. Suppose furthermore that, for any $j, n \geq 1$, there exists a sequence r(j) with

$$r_n^2(j) \le r(j) , \qquad \sum_{j=1}^{\infty} r(j) < \infty.$$
 (3.9)

Then we have

$$\sqrt{n} \left(V(p_1^j, \dots, p_k^j)_t^n - \rho_{p_1^j, \dots, p_k^j}^{(n)} t \right)_{1 \le j \le d} \Rightarrow \beta^{1/2} B_t , \qquad (3.10)$$

where B is a d-dimensional Brownian, β is a d × d-dimensional matrix given by

$$\beta_{ij} = \lim_{n \to \infty} n \ cov \Big(V_Q(p_1^i, \dots, p_k^i)_1^n, V_Q(p_1^j, \dots, p_k^j)_1^n \Big) \ , \qquad 1 \le i, j \le d \ , \tag{3.11}$$

and the convergence holds in the space $\mathcal{D}([0,T]^d)$ equipped with the uniform topology.

Proof of Theorem 2: See Section 7.

Remark 1 It is possible that the condition (3.8) in its present form is not required. However, it leads to an explicit formula for the asymptotic variance in the central limit theorem. In order to allow substitution of $\rho_{p_1^j,...,p_k^j}^{(n)}$ by $\rho_{p_1^j,...,p_k^j}$ in (3.10) the requirement (3.8) must be strengthened so as to ensure that $\sqrt{n} \left(\rho_{p_1^j,...,p_k^j}^{(n)} - \rho_{p_1^j,...,p_k^j} \right) \to 0$. For further discussion of this, see Remark 13 in Section 4 and Section 5.3. Remark 2 Similar asymptotic results can be obtained for general quantities of the form

$$\frac{1}{n} \sum_{i=1}^{[nt]-k+1} H(X_{i,n}, \dots, X_{i+k-1,n})$$
(3.12)

for some function $H : \mathbb{R}^k \to \mathbb{R}$. Let *m* denote the Hermite index of *H* (notice that the Hermite index of the power function used in (3.2) is 2). Replace the condition (3.3) by

$$|r_n^m(j)| \le r(j) , \qquad \frac{1}{n} \sum_{j=1}^{n-1} r(j) \to 0$$

and (3.9) by

$$|r_n^m(j)| \le r(j)$$
, $\sum_{j=1}^{\infty} r(j) < \infty$.

Then Theorem 1 and 2 hold true for the functional (3.12) provided that $EH^2(N_k(0,\Sigma)) < \infty$ for any invertible $\Sigma \in \mathbb{R}^{k \times k}$. We omit the details.

Remark 3 Ho and Sun [HoSu87] have shown a non-functional version of Theorem 2 for statistics of the type (3.12) when the correlation function r_n does not depend on n. To the best of our knowledge Theorem 2 is the first central limit theorem for (general) multipower variation of a row-wise stationary Gaussian process.

4 Multipower variation for BSS processes

Armed with the general theorems proved in Section 3 we are now set to establish laws of large numbers and central limit results for multipower variations in the framework of the Browninan semistationary processes. The regularity conditions invoked are given in a first subsection, while the next states the theorems, the main parts of the proofs being postponed to Section 7; the third subsection discusses the nature of the, rather technical, regularity conditions and describes a set of simpler assumptions that are more amenable to checking.

4.1 Conditions

We consider a filtered probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t \geq 0}, P)$, assuming the existence thereon of a \mathcal{BSS} process, for the time being without drift term, i.e.

$$Y_t = \int_{-\infty}^t g(t-s)\sigma_s W(\mathrm{d}s) \tag{4.1}$$

where W is an \mathcal{F}_t -stochastic Wiener measure, σ is an \mathcal{F}_t -adapted and càdlàg volatility process and $g: (0,\infty) \to \mathbb{R}$ is a deterministic continuous memory function with $g \in L^2((0,\infty))$. We also require $\int_{-\infty}^t g^2(t-s)\sigma_s^2 ds < \infty$ a.s. to ensure that $Y_t < \infty$ a.s. for all $t \ge 0$. By an \mathcal{F}_t -stochastic Wiener measure we understand a Gaussian stochastic measure such that, for any Borelian set A with $E(W(A)^2) < \infty$

$$W(A) \sim N(0, m(A)),$$

where *m* is the Lebesgue measure, and if $A \subseteq [t, +\infty)$ then W(A) is independent of \mathcal{F}_t . Note that $\left\{B_t := \int_a^t W(\mathrm{d}s), t \ge a\right\}$ is a standard Brownian motion starting in *a*.

The process Y is assumed to be observed at time points $t_i = i/n$, i = 1, ..., [nt]. Now, let G be the stationary Gaussian process defined as

$$G_t = \int_{-\infty}^t g(t-s)W(\mathrm{d}s). \tag{4.2}$$

This is an important auxiliary object in the study of \mathcal{BSS} processes. Note that G belongs to the type of processes ocurring in (2.5), and that the autocorrelation function of G is

$$r(t) = \int_0^\infty \bar{g}(t+u)\bar{g}(u)du, \qquad (4.3)$$

where $\bar{g}(t) = g(t)/||g||$ with ||g|| the L^2 norm of g. We are interested in the asymptotic behaviour of the functionals

$$V(Y, p_1, \dots, p_k)_t^n = \frac{1}{n\tau_n^{p_+}} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n Y|^{p_j} , \qquad p_1, \dots, p_k \ge 0 ,$$

where $\Delta_i^n Y = Y_{\frac{i}{n}} - Y_{\frac{i-1}{n}}$ and $\tau_n^2 = \overline{R}(1/n)$ with $\overline{R}(t) = E[|G_{s+t} - G_s|^2]$, $t \ge 0$. In the following we assume that the function g is continuously differentiable on $(0, \infty)$, g' is non-increasing on (b, ∞) for some b > 0 and $g' \in L^2((\varepsilon, \infty))$ for any $\varepsilon > 0$. Moreover, we assume that for any t > 0

$$F_t = \int_1^\infty (g'(s))^2 \sigma_{t-s}^2 ds < \infty \tag{4.4}$$

almost surely. We shall extend the domain of g to \mathbb{R} by taking g(x) = 0, for $x \leq 0$.

Remark 4 The assumption (4.4) ensures that the process Y has the same "smoothness" as the process G (see Lemma 1 in Section 7). It is rather easy to check in practice, because it is implied by the condition $EF_t < \infty$ for t > 0. Furthermore, if g has bounded support assumption (4.4) is trivially fulfilled since σ is càdlàg.

Remark 5 Let us note again that the process Y is, in general, not a semimartingale. In particular, this is the case when $g' \notin L^2((0,\infty))$. For a closer discussion, see [BNSch09]. On the other hand, the process Y is not of the form (2.5). Thus, we require new methods to prove the asymptotic results for $V(Y, p_1, \ldots, p_k)_t^n$. Processes of the form (4.1) are used for modelling velocity of turbulent flows, see [BNSch07], [BNSch08a], [BNSch08b]. The function g, that is used in such models, behaves often as t^{δ} near the origin. Hence, when $\delta \in (-1/2, 1/2) \setminus \{0\}$, Y is neither a differentiable process nor a semimartingale (because $g' \notin L^2((0,\infty))$). This is the primary case of our interest. We define the correlation function of the increments of G:

$$r_n(j) = \cos\left(\frac{\Delta_1^n G}{\tau_n}, \frac{\Delta_{1+j}^n G}{\tau_n}\right) = \frac{\bar{R}(\frac{j+1}{n}) + \bar{R}(\frac{j-1}{n}) - 2\bar{R}(\frac{j}{n})}{2\tau_n^2} , \qquad j \ge 0$$

Next, we introduce a class of measures that is crucial for our purposes. We define (recall that g(x) := 0 for $x \le 0$)

$$\pi^{n}(A) = \frac{\int_{A} (g(x - \frac{1}{n}) - g(x))^{2} \mathrm{d}x}{\int_{0}^{\infty} (g(x - \frac{1}{n}) - g(x))^{2} \mathrm{d}x} , \qquad A \in \mathcal{B}(\mathbb{R}).$$
(4.5)

We further set $\overline{\pi}^n(x) = \pi^n(\{y: y > x\})$. Note that π^n is a probability measure on \mathbb{R}_+ .

For the weak law of large numbers we require the following assumptions:

(LLN): There exists a sequence r(j) with

$$r_n^2(j) \le r(j)$$
, $\frac{1}{n} \sum_{j=1}^{n-1} r(j) \to 0.$

Moreover, it holds that

$$\lim_{n \to \infty} \overline{\pi}^n(\varepsilon) = 0 \tag{4.6}$$

for any $\varepsilon > 0$.

Remark 6 (i) The first condition of (LLN) is adapted from Theorem 1. It guarantees the ucp convergence of $V(G, p_1, \ldots, p_k)_t^n$. The second condition of (LLN) says that the whole mass of the measure π^n concentrates at 0. In particular, it is equivalent to the weak convergence

$$\pi^n \to \delta_0$$
,

where δ_0 is the Dirac measure at 0.

(ii) The condition (4.6) is absolutely crucial for the limit theorems given in the next subsection. When this condition is violated things become more complicated. In particular, it may lead to a different stochastic limit of V(Y, p₁,..., p_k)ⁿ_t (see an example in Section 5). Intuitively, this can be explained by the observation that the increments Δⁿ_iY contain substantial information about the volatility (far) outside of the interval [i-1/n, i/n] when the condition (4.6) does not hold. Thus, in general, we can not expect the limit described in Theorem 3 below.

Now, we introduce the assumptions for the central limit theorem:

(CLT): Assumption (LLN) holds, and

$$r_n(j) \to \rho(j) , \qquad j \ge 0,$$

where $\rho(j)$ is the correlation function of $(Q_i)_{i\geq 1}$, as introduced in (3.6). Furthermore, there exists a sequence r(j) such that, for any $j, n \geq 1$,

$$r_n^2(j) \le r(j)$$
, $\sum_{j=1}^{\infty} r(j) < \infty$,

and for some $\gamma \in (0, 1]$ we have

$$E[|\sigma_t - \sigma_s|^2] \le C|t - s|^{2\gamma}.$$
(4.7)

Finally, set $p = \min_{1 \le i \le k, 1 \le j \le d}(p_i^j)$. Assume that $\gamma(p \land 1) > \frac{1}{2}$ and that there exists a constant $\lambda < -\frac{1}{p \land 1}$ such that for any $\varepsilon_n = O(n^{-\kappa})$, $\kappa \in (0, 1)$, we have

$$\overline{\pi}^n(\varepsilon_n) = O(n^{\lambda(1-\kappa)}). \tag{4.8}$$

Remark 7 Note that if σ is stationary with increasing complementary autocorrelation function $\bar{\omega}$, say, then condition (4.7) reduces to $\bar{\omega} < Cu^{2\gamma}$. In particular, for the realized quadratic variation we require $1/2 < \gamma \leq 1$ meaning that $\varpi(u)$ should go to 0 quite fast or, equivalently, that $\omega(u)$ tends fast to 1. In other words, σ should not vary too fast.

Remark 8 Assumption (4.7) is only one of a variety of possible regularity conditions on σ that can lead to a central limit theorem for multipower variations of the kind we are after, and it is some way away from being necessary. For instance, Theorem 5 below will also hold is σ is a sum of two processes, one of which satisfies (4.7) and the other having the property that (almost) every sample path is of bounded variation on finite intervals.

Remark 9 The first part of assumption (CLT) ensures the weak convergence of the standardized version of $V(G, p_1, \ldots, p_k)_t^n$. The condition (4.8) is certainly stronger than (4.6) in (LLN). In Section 4.3 we will explain how (4.8) can be checked in practice.

4.2 Limit theorems

In this section we present the limit laws of multipower variations of \mathcal{BSS} processes, in part widening the scope slightly to allow more general drift terms. Recall that the (realised) multipower variation of a process Y of the form (4.1) is defined as

$$V(Y, p_1, \dots, p_k)_t^n = \frac{1}{n\tau_n^{p_+}} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n Y|^{p_j} , \qquad p_1, \dots, p_k \ge 0 , \qquad (4.9)$$

where $\tau_n^2 = \bar{R}\left(\frac{1}{n}\right)$ and $p_+ = \sum_{j=1}^k p_j$. Our first result is the following probability limit theorem.

Theorem 3 Assume that the condition (LLN) holds with Y given by (4.1). Define

$$\rho_{p_1,\dots,p_k}^{(n)} = E\left[\left|\frac{\Delta_1^n G}{\tau_n}\right|^{p_1} \cdots \left|\frac{\Delta_k^n G}{\tau_n}\right|^{p_k}\right].$$

Then we have

$$V(Y, p_1, \dots, p_k)_t^n - \rho_{p_1, \dots, p_k}^{(n)} \int_0^t |\sigma_s|^{p_+} \mathrm{d}s \xrightarrow{ucp} 0.$$
(4.10)

Proof of Theorem 3: See section 7.

Next, we demonstrate that the functional $V(Y, p_1, \ldots, p_k)_t^n$ is robust to some types of drift processes, which includes the kind of drift term occurring in \mathcal{BSS} processes.

Proposition 4 Consider a process $Z = Z_1 + Z_2$, where $Z_2 = Y$ is given by (4.1). Assume that the condition (LLN) holds and define

$$V(Z_1, Z_2; p_1, \dots, p_k; \iota_1, \dots, \iota_k)_t^n = \frac{1}{n\tau_n^{p_+}} \sum_{i=1}^{[nt]-k+1} \prod_{j=1}^k |\Delta_{i+j-1}^n Z_{\iota_j}|^{p_j} , \qquad p_1, \dots, p_k \ge 0 ,$$

where $\iota_1, \ldots, \iota_k \in \{1, 2\}$. If for any t > 0 and any $(\iota_1, \ldots, \iota_k) \neq (2, \ldots, 2)$,

$$V(Z_1, Z_2; p_1, \dots, p_k; \iota_1, \dots, \iota_k)_t^n \xrightarrow{P} 0 , \qquad (4.11)$$

then

$$V(Z, p_1, \ldots, p_k)_t^n - \rho_{p_1, \ldots, p_k}^{(n)} \int_0^t |\sigma_s|^{p_+} \mathrm{d}s \xrightarrow{ucp} 0.$$

Proof of Proposition 4: For simplicity we consider the case k = 2. Since $V(Z, p_1, p_2)_t^n$ is increasing in t and the process $\rho_{p_1,p_2}^{(n)}t$ is continuous in t, it is sufficient to prove $V(Z, p_1, p_2)_t^n - \rho_{p_1,p_2}^{(n)}t \xrightarrow{P} 0$ for a fixed t > 0.

Assume first that $0 \le p_1, p_2 \le 1$. We have $||x_1+y_1|^{p_1}|x_2+y_2|^{p_2}-|y_1|^{p_1}|y_2|^{p_2}| \le C(|x_1|^{p_1}|x_2|^{p_2}+|x_1|^{p_1}|y_2|^{p_2}+|y_1|^{p_1}|x_2|^{p_2})$ (here and elsewhere *C* denotes a constant the value of which may change from line to line). Hence we deduce

$$|V(Z, p_1, p_2)_t^n - V(Y, p_1, p_2)_t^n|$$

$$\leq C(V(Z_1, Z_2; p_1, p_2; 1, 1)_t^n + V(Z_1, Z_2; p_1, p_2; 1, 2)_t^n + V(Z_1, Z_2; p_1, p_2; 2, 1)_t^n),$$

and we obtain Proposition 4 by (4.11).

Next, assume that $p_1 \leq p_2, p_2 > 1$. We deduce that

$$|(V(Z, p_1, p_2)_t^n)^{1/p_2} - (V(Y, p_1, p_2)_t^n)^{1/p_2}|$$

 $\leq C \Big((V(Z_1, Z_2; p_1, p_2; 1, 1)_t^n)^{1/p_2} + (V(Z_1, Z_2; p_1, p_2; 1, 2)_t^n)^{1/p_2} + (V(Z_1, Z_2; p_1, p_2; 2, 1)_t^n)^{1/p_2} \Big) ,$

which completes the proof of Proposition 4.

Remark 10 The multipower variation is robust to drift processes Z_1 that are smoother than the process Y. Assume, for instance, that the process Z_1 satisfies

$$E[|Z_1(t) - Z_1(s)|^p] = o(\bar{R}^{p/2}(|t-s|))$$

for every p > 0. In this case condition (4.11) is obviously satisfied.

Next, we demonstrate a joint central limit theorem for a family $(V(Y, p_1^j, \ldots, p_k^j)_t^n)_{1 \le j \le d}$ of multipower variations. Let \mathcal{G} be the σ -algebra generated by the auxiliary process G.

Theorem 5 Assume that the process σ is \mathcal{G} -measurable and the condition (CLT) holds. Then we obtain the stable convergence

$$\sqrt{n} \Big(V(Y, p_1^j, \dots, p_k^j)_t^n - \rho_{p_1^j, \dots, p_k^j}^{(n)} \int_0^t |\sigma_s|^{p_+^j} \mathrm{d}s \Big)_{1 \le j \le d} \xrightarrow{\mathcal{G}-st} \int_0^t A_s^{1/2} \mathrm{d}B_s , \qquad (4.12)$$

where B is a d-dimensional Brownian motion that is defined on an extension of the filtered probability space $(\Omega, \mathcal{F}, (\mathcal{F}_t)_{t\geq 0}, P)$ and is independent of \mathcal{F} , and A is a $d \times d$ -dimensional process given by

$$A_s^{ij} = \beta_{ij} |\sigma_s|^{p_+^i + p_+^j} , \qquad 1 \le i, j \le d , \qquad (4.13)$$

and the $d \times d$ matrix β is defined in (3.11).

Proof of Theorem 5: See Section 7.

Remark 11 We require the \mathcal{G} -measurability of the process σ for the following reason. In fact, in Section 7, we will first prove the joint weak convergence

$$\left(G_t, \sqrt{n}\left(V(G, p_1^j, \dots, p_k^j)_t^n - \rho_{p_1^j, \dots, p_k^j}^{(n)} t\right)_{1 \le j \le d}\right) \Rightarrow \left(G_t, \beta^{1/2} B_t\right).$$

This implies the joint convergence

$$\left(\sigma_{t_l}, \sqrt{n}\left(V(G, p_1^j, \dots, p_k^j)_{t_l}^n - \rho_{p_1^j, \dots, p_k^j}^{(n)} t_l\right)_{1 \le j \le d}\right)_{1 \le l \le m} \Rightarrow \left(\sigma_{t_l}, \beta^{1/2} B_{t_l}\right)_{1 \le l \le m},$$

for any fixed m, because σ is \mathcal{G} -measurable. From the latter we deduce the \mathcal{G} -stable convergence of Theorem 5 by a certain approximation technique.

From the above argument it is easy to see that Theorem 5 remains true when σ is independent of W. Hence, Theorem 5 holds for any process $\sigma = \sigma^{(1)} + \sigma^{(2)}$ such that $\sigma^{(1)}$ is \mathcal{G} -measurable, $\sigma^{(2)}$ is independent of W and both processes $\sigma^{(1)}$ and $\sigma^{(2)}$ satisfy the conditions of assumption (CLT).

For completeness we provide a condition under which the above central limit theorem is robust to a potential drift process.

Proposition 6 Consider a process $Z = Z_1 + Z_2$, where $Z_2 = Y$ is given by (4.1). Assume that the conditions of Theorem 5 hold and d = 1. If

$$\sqrt{n}V(Z_1, Z_2; p_1, \dots, p_k; \iota_1, \dots, \iota_k)_t^n \xrightarrow{P} 0 , \qquad (4.14)$$

for any t > 0 and any $(\iota_1, \ldots, \iota_k) \neq (2, \ldots, 2)$, then we obtain

$$\sqrt{n}\Big(V(Z,p_1,\ldots,p_k)_t^n - \rho_{p_1,\ldots,p_k}^{(n)} \int_0^t |\sigma_s|^{p_+} \mathrm{d}s\Big)_{1 \le j \le d} \xrightarrow{\mathcal{G}-st} \sqrt{\beta} \int_0^t |\sigma_s|^{p_+} \mathrm{d}B_s.$$

Proof of Proposition 6: Proposition 6 can be proved by the same methods as Proposition 4 (details omitted). \Box

Clearly, Proposition 6 extends in a direct manner to the multivariate setting $(d \ge 1)$. Concerning the possibility of substituting $\rho_{p_1^j,\ldots,p_k^j}^{(n)}$ by $\rho_{p_1^j,\ldots,p_k^j}$ in the above conclusions, see Remark 1.

4.3 Discussion of assumptions

We start our discussion again by considering the auxiliary, centered stationary Gaussian, process

$$G_t = \int_{-\infty}^t g(t-s)W(\mathrm{d}s).$$

First of all, we want to demonstrate how Theorems 1 and 2 apply for the multipower variation of the process G. In other words, we will give a hint how to check the conditions of these theorems.

Recall the definition (2.7) of the variance function R of the increments of G and note that

$$\bar{R}(t) = E[|G_{s+t} - G_s|^2] = \int_0^t g^2(x) dx + \int_0^\infty (g(t+x) - g(x))^2 dx , \qquad t \ge 0.$$

Clearly, the asymptotic behaviour of the multipower variation of the process G is fully determined by the behaviour of the function \overline{R} near 0. As we deal with a continuous process G, it is natural to assume that $\overline{R}(t)$ behaves essentially as t^{α} (for some $\alpha > 0$) near 0 (later on we will formalize this assumption). Since the case where the paths of G are differentiable (a.s.) is not very interesting for us (because the consistency can be deduced by the mean value theorem), we concentrate on the region $0 < \alpha < 2$ (the corresponding g(t) behaving as $t^{(\alpha-1)/2}$).

Let us introduce a new set of assumptions that correspond to the previous discussion. These assumptions were proposed by Guyon and Leon in [GuyLe89] (those authors considered the case of centered stationary Gaussian processes X; this relates to the \mathcal{BSS} setting with σ constant) and the same assumptions were used in [BNCP09] and [BNCPW09].

(A1) $\overline{R}(t) = t^{\alpha}L_0(t)$ for some $\alpha \in (0,2)$ and some positive slowly varying (at 0) function L_0 , which is continuous on $(0,\infty)$.

(A2) $\bar{R}''(t) = t^{\alpha-2}L_2(t)$ for some slowly varying function L_2 , which is continuous on $(0,\infty)$.

(A3) There exists $b \in (0, 1)$ with

$$K = \limsup_{x \to 0} \sup_{y \in [x, x^b]} \left| \frac{L_2(y)}{L_0(x)} \right| < \infty.$$

Recall that a function $L: (0, \infty) \to \mathbb{R}$ is called slowly varying at 0 when the identity

$$\lim_{x \searrow 0} \frac{L(tx)}{L(x)} = 1$$

holds for any fixed t > 0.

Now, note that under assumption (A1) we have, for any $j \ge 1$,

$$r_{n}(j) = \operatorname{cov}\left(\frac{\Delta_{1}^{n}G}{\tau_{n}}, \frac{\Delta_{1+j}^{n}G}{\tau_{n}}\right)$$
$$= \frac{\bar{R}(\frac{j+1}{n}) + \bar{R}(\frac{j-1}{n}) - 2\bar{R}(\frac{j}{n})}{2\bar{R}(\frac{1}{n})} \to \rho(j) = \frac{1}{2}((j+1)^{\alpha} - 2j^{\alpha} + (j-1)^{\alpha}), \quad (4.15)$$

because L_0 is slowly varying at 0. It is obvious that $\rho(j)$ is the correlation function of the discrete time stationary Gaussian process $Q_i = B_i^{\alpha/2} - B_{i-1}^{\alpha/2}$, where $B^{\alpha/2}$ is a fractional Brownian motion with parameter $\alpha/2$.

As shown in [GuyLe89] and [BNCP09] assumptions (A1)-(A3) imply that for any $\alpha \in (0, 2)$ there exists a number $h = h(\alpha) > 0$ such that

$$r_n^2(j) \le r(j) = Cj^{-h}$$

for all $j \ge 1$ and all, but finitely many, n. Hence, under (A1)-(A3), the condition of Theorem 1 is satisfied for all $\alpha \in (0,2)$, because $\frac{1}{n} \sum_{j=1}^{n} r(j) \to 0$.

Moreover, the conditions (A1)-(A3) imply that for any $\alpha \in (0, 3/2)$ there exists a number $\bar{h} = \bar{h}(\alpha) > 1$ such that

$$r_n^2(j) \le r(j) = Cj^{-\bar{h}-1}$$

for all $j \ge 1$ and all, but finitely many, n. Consequently, when $\alpha \in (0, 3/2)$ and (A1)-(A3) hold, the conditions of Theorem 2 are satisfied with $Q_i = B_i^{\alpha/2} - B_{i-1}^{\alpha/2}$, $i \ge 1$, since $\sum_{j=1}^{\infty} r(j) < \infty$.

Remark 12 It is easy to see that $\frac{\bar{R}(\frac{j+1}{n}) + \bar{R}(\frac{j-1}{n}) - 2\bar{R}(\frac{j}{n})}{2\bar{R}(\frac{1}{n})} \to \rho(j)$ implies that for all $j \ge 1$ there exits an s(j) such that

$$\frac{R(\frac{j}{n})}{\bar{R}(\frac{1}{n})} \to s(j).$$

Since the result in Theorem 5 is independent of the scale of time we use, we must have

$$\frac{\bar{R}(j\Delta)}{\bar{R}(\Delta)} \underset{\Delta\downarrow 0}{\rightarrow} s(j),$$

for any Δ and then s(jk) = s(j)s(k); consequently $s(j) = j^{\alpha}$, for a certain $\alpha \in \mathbb{R}$. Moreover, since $(j+1)^{\alpha} - 2j^{\alpha} + (j-1)^{\alpha}$ is a covariance function we have $0 < \alpha < 2$. So in the present setting $(Q_i)_{i\geq 1}$, as defined in Section 3, is always a standard fractional Gaussian noise. **Remark 13** In Theorem 5 the quantity $\rho_{p_1,\dots,p_k}^{(n)}$ can be replaced by its limit ρ_{p_1,\dots,p_k} (which necessarily exists because $r_n(j) \to \rho(j)$ for all $j \ge 1$) whenever

$$\sqrt{n}(\rho_{p_1,\dots,p_k}^{(n)} - \rho_{p_1,\dots,p_k}) \to 0.$$

As the quantity $\rho_{p_1,\ldots,p_k}^{(n)}$ is a continuously differentiable function of $r_n(1),\ldots,r_n(k-1)$ (recall that we are in the non-degenerate case) the latter follows from

$$\sqrt{n} \max_{j=1,\dots,k-1} |r_n(j) - \rho(j)| \to 0.$$

To study the above convergence let us introduce the notion of the second order regular variation (however, we restrict ourselves to the second order regular variation of slowly varying functions). A slowly varying function $L : (0, \infty) \to \mathbb{R}$ is called second order regular varying (at 0) with parameter $\gamma > 0$ if there exists a function $A : (0, \infty) \to \mathbb{R}$, positive or negative, with $\lim_{x\to 0} A(x) = 0$ such that

$$\lim_{x \to 0} \frac{\frac{L(tx)}{L(x)} - 1}{A(x)} = \frac{t^{\gamma} - 1}{\gamma}$$

holds for all t > 0. It is known that if the limit on the left-hand side exists for all t > 0 it must be essentially of the form $\frac{t^{\gamma}-1}{\gamma}$. Furthermore, if L is second order regular varying with parameter $\gamma > 0$ then the function A must be regular varying with parameter γ , i.e.

$$\lim_{x \to 0} \frac{A(tx)}{A(x)} = t^{\gamma}$$

for all t > 0. Notice that the parameter γ essentially gives the rate of convergence for $\frac{L(tx)}{L(x)} \rightarrow 1$. Observing the convergence in (4.15) we deduce that

$$\sqrt{n} \max_{j=1,\dots,k-1} |r_n(j) - \rho(j)| \to 0$$

holds when the slowly varying function L_0 , which appears in the assumption (A1), is second order regular varying with parameter $\gamma > \frac{1}{2}$.

Now, let us see what the conditions (A1)-(A3) mean for the memory function g. For simplicity let us consider functions of the form

$$g(x) = x^{\delta} \mathbf{1}_{(0,1]}(x) , \qquad x > 0.$$
 (4.16)

For such functions we readily obtain assumptions (A1)-(A2) with

$$\alpha = 2\delta + 1$$
, $\delta \in (-\frac{1}{2}, 0) \cup (0, \frac{1}{2}).$

(the technical assumption (A3) has to be checked separately; for an example see Section 5). Note that for $\delta = 0$, for which assumption (A2) does not hold, the process G is a semimartingale and the multipower variations can be treated as in [BNGJPS06]. Next, we discuss the assumptions of subsection 4.1 for the function g defined in (4.16). Recall that condition (4.4) is automatically satisfied for functions g with compact support (as in (4.16)). A straightforward calculation shows that

$$\overline{\pi}^n(\varepsilon) = O((n\varepsilon)^{2\delta - 1}).$$

for any $\varepsilon > \frac{1}{n}$. Thus, the condition (4.6) of **(LLN)** is satisfied (because $2\delta - 1 < 0$) and Theorem 3 is valid for all $\delta \in (-1/2, 0) \cup (0, 1/2)$.

Finally, we explain how to verify the condition (4.8) of **(CLT)**. Recall that $p = \min_{1 \le i \le k, 1 \le j \le d} (p_i^j)$. Let $\varepsilon_n = n^{-\kappa}$, $\kappa \in (0, 1)$. We readily deduce that

$$\overline{\pi}^n(\varepsilon) = O(n^{\lambda(1-\kappa)}), \qquad \lambda = 2\delta - 1.$$

Thus, condition (4.8) is satisfied if

$$\lambda < -\frac{1}{1 \wedge p}.$$

We immediately deduce that Theorem 5 holds if

$$\begin{split} p \geq 1: \qquad \gamma > \frac{1}{2}, \quad \delta \in (-\frac{1}{2}, 0) \\ \frac{1}{2} \frac{1}{2p}, \quad \delta \in (-\frac{1}{2}, \frac{p-1}{2p}) \end{split}$$

Remark 14 Clearly, we can deal with a larger class of functions g than $g(x) = x^{\delta} 1_{(0,1]}(x)$. Assume that condition (4.4) holds. In the following we consider functions L_g , $L_{g'}$, which are continuous on $(0, \infty)$ and slowly varying at 0. We assume the following conditions:

Assumption: $g \in L^2((0,\infty))$ and for some $\delta \in (-1/2,0) \cup (0,1/2)$ it holds that

- (i) $g(x) = x^{\delta} L_g(x)$.
- (ii) $g'(x) = x^{\delta-1}L_{g'}(x)$ and, for any $\varepsilon > 0$, $g' \in L^2((\varepsilon, \infty))$. Moreover, g' is non-increasing on (b, ∞) for some b > 0.

We further assume that the function

$$\bar{R}(t) = \int_0^t g^2(x) dx + \int_0^\infty (g(t+x) - g(x))^2 dx$$

satisfies the conditions (A1)-(A3) with $\alpha = 2\delta + 1$.

Under these assumptions we conclude (as for the simple example $g(x) = x^{\delta} 1_{(0,1]}(x)$) that Theorem 3 holds for any $\delta \in (-1/2, 0) \cup (0, 1/2)$, and Theorem 5 holds when further

$$p \ge 1: \qquad \gamma > \frac{1}{2}, \quad \delta \in (-\frac{1}{2}, 0)$$

$$\frac{1}{2} \frac{1}{2p}, \quad \delta \in (-\frac{1}{2}, \frac{p-1}{2p}).$$

In both cases we have $Q_i = B_i^{\delta + \frac{1}{2}} - B_{i-1}^{\delta + \frac{1}{2}}, i \ge 1.$

5 Examples

The present Section discusses two examples of choice of the damping function g and the associated probabilistic limit behaviour.

As above let r denote the autocorrelation function of $G = \int_{-\infty}^{\cdot} g(\cdot - s)W(ds)$, and we write \bar{r} for 1 - r. Note that assumptions **(A1)-(A3)** could equivalently have been formulated in terms of \bar{r} rather than \bar{R} (since $\bar{R}(t) = 2 ||g||^2 \bar{r}(t)$).

Suppose first that

$$g(t) = e^{-\lambda t} \mathbf{1}_{(0,1)}(t)$$

with $\lambda > 0$. This example (for a detailed discussion see [BNSch09]) is a non-semimartingale case, and it can be shown that $\pi^n \to \pi$, with π given by

$$\pi = \frac{1}{1 + e^{-2\lambda}}\delta_0 + \frac{1}{1 + e^{2\lambda}}\delta_1,$$

where δ_x is the Dirac measure at x. Moreover,

$$V(Y,2)_t^n \xrightarrow{P} \sigma_t^{2+} - \left(1 + e^{2\lambda}\right)^{-1} \sigma_{-t}^{2+}.$$

Thus, in particular, we do not have $V(Y,2)_t^n \xrightarrow{P} \sigma_t^{2+}$. Note that in this example assumption (A2) is not satisfied.

Our main example is

$$g(t) = t^{\nu - 1} e^{-\lambda t} \mathbf{1}_{(0,\infty)}(t)$$
(5.1)

for $\lambda > 0$ and with $\nu > \frac{1}{2}$. (So, for t near 0, g(t) behaves as t^{δ} with $\delta = \nu - 1$.) The following two subsections discuss the properties of the autocorrelation function r for this g, presenting exact formulae (in terms of the Bessel functions K_{ν}) for r and its derivatives in subsection 5.1 and deriving asymptotic properties of $\bar{r}(t) = 1 - r(t)$ for $t \to 0$ in subsection 5.2. Armed with these results we show in subsection 5.3 that assumptions **(A1)-(A3)** are met provided $\alpha = 2\nu - 1 \in (0, 2)$, i.e. $\nu \in (\frac{1}{2}, \frac{3}{2})$ and that $\rho_{p_1^j, \dots, p_k^j}^{(n)}$ may be substituted by $\rho_{p_1^j, \dots, p_k^j}$ in the central limit theory provided $\nu \in (\frac{1}{2}, \frac{5}{4})$.

Remark 15 The derivative g' of g is not square integrable if $\frac{1}{2} < \nu < 1$ or $1 < \nu \leq \frac{3}{2}$; hence, in these cases Y is not a semimartingale. For $\frac{1}{2} < \nu < 1$ we have $g(0+) = \infty$ while g(0+) = 0 when $1 < \nu \leq \frac{3}{2}$. Of course, for $\nu = 1$ the process $Y = \int_{-\infty}^{\cdot} g(\cdot - s)\sigma_s W(ds)$ is simply a modulated version of the Gaussian Ornstein-Uhlenbeck process, and in particular, a semimartingale. Note also that when $\nu > \frac{3}{2}$ then Y is of finite variation and hence, trivially, a semimartingale.

Remark 16 With $Y = \int_{-\infty}^{\cdot} g(\cdot - s)\sigma_s W(ds)$, suppose that the volatility process is constant, $\sigma_t = \sigma$. This is a special case of the class of stationary Gaussian processes discussed by

[GuyLe89]. These authors showed that, under conditions (A1)-(A3),

$$\frac{(2\lambda)^{2\nu-1}}{\Gamma(2\nu-1)}\frac{1}{2n\tau_n}V(Y,2)_t^n \xrightarrow{p} \sigma^2$$

More generally, they derived associated (nonfeasible) limit law results which in the present example (of (5.1) (not considered by [GuyLe89]), implies that the limit distribution is normal if $\frac{1}{2} < \nu < \frac{5}{4}$, with rate $n^{3/2}\bar{r}\left(\frac{1}{n}\right)$, while for $\frac{5}{4} < \nu < \frac{3}{2}$ it belongs to the second order chaos, the rate of convergence being then $n^{-2\nu+3}$. Extension to the power variations $V(Y,q)_t^n$, q > 0, are also given in [GuyLe89]. In [BNCP09] extensions of the normal limit results to processes of the form $\int_0^{\cdot} \sigma_s dG_s$ where G is a Gaussian process with stationary increments are established. The earlier paper [Guy87] contains a set of sufficient conditions for convergence in probability of normalised versions of $V(Y,q)_t^n$ in cases where the process Y is nonstationary Gaussian. The conditions are rather restrictive; in particular, they can only apply to certain types of processes for which $\bar{r}\left(\frac{1}{n}\right)$ behaves as a constant times n^{-1} as $n \to \infty$.

For the following analysis we need a number of, mostly well known, properties of modified Bessel functions of the third type K_{ν} . These are given in the Appendix.

5.1 Formulae for r and its derivatives

With g given by (5.1) the autocorrelation function r of $G = \int_{-\infty}^{\cdot} g(\cdot - s)W(ds)$ has the form (cf. formula (4.3))

$$r(t) = \frac{(2\lambda)^{2\nu-1}}{\Gamma(2\nu-1)} e^{-\lambda t} \int_0^\infty (t+u)^{\nu-1} u^{\nu-1} e^{-2\lambda u} \mathrm{d}u.$$
(5.2)

Hence by formulae (A.1.4), (A.1.5) and (A.1.10) in the Appendix we find

$$\begin{aligned} \|g\|^{2} r(t) &= \sqrt{\frac{2}{\pi}} \Gamma(\nu) \, 2^{-\nu} \lambda^{-\nu+\frac{1}{2}} t^{\nu-\frac{1}{2}} K_{\nu-\frac{1}{2}}(\lambda t) \\ &= \sqrt{\frac{2}{\pi}} \Gamma(\nu) \, 2^{-\nu} \lambda^{-2\nu+1} \bar{K}_{\nu-\frac{1}{2}}(\lambda t) \\ &= \frac{1}{2} \frac{\Gamma(\nu) \, \Gamma(\nu-\frac{1}{2})}{\Gamma(\frac{1}{2})} \lambda^{-2\nu+1} \check{K}_{\nu-\frac{1}{2}}(\lambda t) \,. \end{aligned}$$
(5.3)

where we have used the doubling formula

$$\Gamma(2\nu - 1) = 2^{2\nu - 2} \frac{\Gamma(\nu) \Gamma\left(\nu - \frac{1}{2}\right)}{\Gamma\left(\frac{1}{2}\right)}.$$
(5.4)

It follows, by (A.1.7), that

$$r(t) = \check{K}_{\nu - \frac{1}{2}}(\lambda t) \dots$$
 (5.5)

Now, let

$$a(\nu) = 2^{-\nu+1} \Gamma(\nu)^{-1}$$
.

so that

$$\frac{a\left(\nu\right)}{a\left(\nu+1\right)} = 2\nu\tag{5.6}$$

and

 $\check{K}(x) = a(\nu) \,\bar{K}_v(x) \,.$

Supposing for notational simplicity that $\lambda = 1$ and using (A.1.9) and (A.1.8) we find, for $\nu \in (\frac{1}{2}, \frac{3}{2})$,

$$\begin{aligned} \bar{r}'(t) &= -a\left(\nu - \frac{1}{2}\right)t\bar{K}_{\nu-\frac{3}{2}}(t) \\ &= -a\left(\nu - \frac{1}{2}\right)t^{\nu-\frac{1}{2}}K_{\nu-\frac{3}{2}}(t) \\ &= -a\left(\nu - \frac{1}{2}\right)t^{\nu-\frac{1}{2}}K_{\frac{3}{2}-\nu}(t) \\ &= -a\left(\nu - \frac{1}{2}\right)t^{2\nu-2}\bar{K}_{\frac{3}{2}-\nu}(t) \\ &= -\frac{a\left(\nu - \frac{1}{2}\right)}{a\left(\frac{3}{2}-\nu\right)}t^{2\nu-2}\check{K}_{\frac{3}{2}-\nu}(t) \end{aligned}$$
(5.7)

$$\bar{r}''(t) = a\left(\nu - \frac{1}{2}\right) \left\{ t^2 \bar{K}_{\nu - \frac{5}{2}}(t) - \bar{K}_{\nu - \frac{3}{2}} \right\} = a\left(\nu - \frac{1}{2}\right) \left\{ t^{\nu - \frac{1}{2}} K_{\frac{5}{2} - \nu}(t) - t^{\nu - \frac{3}{2}} K_{\frac{3}{2} - \nu}(t) \right\} = a\left(\nu - \frac{1}{2}\right) t^{2\nu - 3} \left\{ \bar{K}_{\frac{5}{2} - \nu}(t) - \bar{K}_{\frac{3}{2} - \nu}(t) \right\}$$
(5.8)

$$\bar{r}^{\prime\prime\prime}(t) = -a\left(\nu - \frac{1}{2}\right) \left\{ t^{3}\bar{K}_{\nu - \frac{7}{2}}(t) - 3t\bar{K}_{\nu - \frac{5}{2}}(t) \right\} \\
= -a\left(\nu - \frac{1}{2}\right) \left\{ t^{\nu - \frac{1}{2}}K_{\frac{7}{2} - \nu}(t) - 3t^{\nu - \frac{3}{2}}K_{\frac{5}{2} - \nu}(t) \right\} \\
= -a\left(\nu - \frac{1}{2}\right) t^{2\nu - 4} \left\{ \bar{K}_{\frac{7}{2} - \nu}(t) - 3\bar{K}_{\frac{5}{2} - \nu}(t) \right\}.$$
(5.9)

5.2 Behaviour of $\bar{r} = 1 - r$ near 0

From the asymptotic expansions (A.2.4), (A.2.5) and (A.2.6) we find that for $t \to 0$ the complementary autocorrelation function $\bar{r}(t) = 1 - r(t)$ behaves as

Table 1

Remark 17 So for $\frac{3}{2} < \nu \leq 2$ the autocorrelation function is twice differentiable at 0 and consequently Y has continuously differentiable sample paths, while for $\frac{1}{2} < \nu \leq \frac{3}{2}$ the sample paths are Lipschitz of order λ for every $0 < \lambda < \nu - \frac{1}{2}$ (cf. [CL67] Section 9.2).

5.3 Verification of assumptions (A1)-(A3)

This subsection establishes that conditions (A1)-(A3) are satisfied (with $\alpha = 2\nu - 1$ and $\nu \in (\frac{1}{2}, \frac{3}{2})$, i.e. $\alpha \in (0, 2)$). Recall that in those conditions we may substitute \bar{r} for \bar{R} .

On account of (5.5) and Table 1 we find that \bar{r} is of the form

$$\bar{r}(t) = t^{2\nu - 1} L_0\left(t\right)$$

with

$$L_{0}(t) = t^{-2\nu+1} \left(1 - \check{K}_{\nu-\frac{1}{2}}(\lambda t) \right)$$

and

$$L_0(t) \to 2^{-2\nu+1} \frac{\Gamma\left(\frac{3}{2} - \nu\right)}{\Gamma\left(\nu + \frac{1}{2}\right)} \quad \text{for} \quad t \to 0.$$

It follows that L_0 is slowly varying at 0, and hence assumption (A1) is met.

In fact, more is true: L_0 is second order slowly varying To see this, note that

$$\frac{L_0(tx)}{L_0(x)} = t^{-2\nu+1} \frac{1 - K_{\nu-\frac{1}{2}}(\lambda tx)}{1 - K_{\nu-\frac{1}{2}}(\lambda x)}$$

from which we find, using formula (A.2.4),

$$\frac{L_{0}(tx)}{L_{0}(x)} = \frac{2^{-2\nu+1}\frac{\Gamma(\frac{3}{2}-\nu)}{\Gamma(\frac{1}{2}+\nu)}(\lambda x)^{2\nu-1} - \frac{1}{4}\frac{1}{\frac{3}{2}-\nu}(\lambda x)^{2}t^{3-2\nu} + O(x^{2\nu+1})}{2^{-2\nu+1}\frac{\Gamma(\frac{3}{2}-\nu)}{\Gamma(\frac{1}{2}+\nu)}(\lambda x)^{2\nu-1} - \frac{1}{4}\frac{1}{\frac{3}{2}-\nu}(\lambda x)^{2} + O(x^{2\nu+1})}{(\lambda x)^{2} + O(x^{2\nu+1})} = \frac{1 - 2^{2(\nu-1)}\frac{\Gamma(\frac{1}{2}+\nu)}{\Gamma(\frac{3}{2}-\nu)}(\lambda x)^{3-2\nu}\frac{t^{3-2\nu}}{3-2\nu} + O(x^{2})}{1 - 2^{2(\nu-1)}\frac{\Gamma(\frac{1}{2}+\nu)}{\Gamma(\frac{3}{2}-\nu)}(\lambda x)^{3-2\nu} + O(x^{2})}.$$

Thus

$$\frac{L_{0}\left(tx\right) }{L_{0}\left(x\right) }-1=A\left(x\right) \frac{t^{3-2\nu }-1}{3-2\nu }+o\left(A\left(x\right) \right)$$

where

$$A(x) = -2^{-2(\nu-1)} \frac{\Gamma(\frac{1}{2} + \nu)}{\Gamma(\frac{3}{2} - \nu)} (\lambda x)^{3-2\nu}$$

which tends to 0 as $x \to 0$. Hence L_0 is second order regulary varying with parameter $\gamma = 3-2\nu$.

It follows that, for the present example, Remark 13 applies provided $3 - 2\nu > \frac{1}{2}$, i.e. $\frac{1}{2} < \nu < \frac{5}{4}$.

Next, we note that

$$\bar{r}''(t) = t^{2\nu - 3} L_2(t)$$

with

$$L_{2}(t) = a\left(\nu - \frac{1}{2}\right) \left\{ \bar{K}_{\frac{5}{2}-\nu}(t) - \bar{K}_{\frac{3}{2}-\nu}(t) \right\},\$$

where the function a is defined by $a(\nu) = 2^{-\nu+1}\Gamma(\nu)^{-1}$ and L_2 is slowly varying at 0 with

$$L_2(t) \to -2^3 \left(\nu - 1\right) \frac{\Gamma\left(\frac{3}{2} - \nu\right)}{\Gamma\left(\nu - \frac{1}{2}\right)} \quad \text{for} \quad t \to 0.$$

The latter follows from the rewrite

$$\begin{split} \bar{r}''(t) &= t^{2\nu-3} a\left(\nu - \frac{1}{2}\right) a\left(\frac{3}{2} - \nu\right)^{-1} \left\{ \left(3 - 2\nu\right) \check{K}_{\frac{5}{2} - \nu}(t) - \check{K}_{\frac{3}{2} - \nu}(t) \right\} \\ &= t^{2\nu-3} 2^2 \frac{\Gamma\left(\frac{3}{2} - \nu\right)}{\Gamma\left(\nu - \frac{1}{2}\right)} \left\{ \left(3 - 2\nu\right) \check{K}_{\frac{5}{2} - \nu}(t) - \check{K}_{\frac{3}{2} - \nu}(t) \right\}. \end{split}$$

Thus (A2) holds.

Finally, we find

$$\begin{split} L_{2}'(t) &= a\left(\nu - \frac{1}{2}\right)\left\{\bar{K}_{\frac{5}{2}-\nu}'(t) - \bar{K}_{\frac{3}{2}-\nu}'(t)\right\} \\ &= a\left(\nu - \frac{1}{2}\right)t\left\{\bar{K}_{\frac{1}{2}-\nu}(t) - \bar{K}_{\frac{3}{2}-\nu}(t)\right\} \\ &= a\left(\nu - \frac{1}{2}\right)t\left\{t^{\frac{1}{2}-\nu}K_{\nu-\frac{1}{2}}(t) - \bar{K}_{\frac{3}{2}-\nu}(t)\right\} \\ &= a\left(\nu - \frac{1}{2}\right)t\left\{t^{-2\nu+1}\bar{K}_{\nu-\frac{1}{2}}(t) - \bar{K}_{\frac{3}{2}-\nu}(t)\right\} \\ &= a\left(\nu - \frac{1}{2}\right)t^{-2\nu+2}\left\{a(\nu - \frac{1}{2})^{-1}t^{-2\nu+1}\check{K}_{\nu-\frac{1}{2}}(t) - a(\nu - \frac{3}{2})^{-1}t^{2\nu-1}\check{K}_{\frac{3}{2}-\nu}(t)\right\} \end{split}$$

Hence (for $\nu \in \left(\frac{1}{2}, \frac{3}{2}\right)$) $L_2(t)$ is increasing near 0. Consequently

$$\limsup_{x \to 0} \sup_{y \in [x, x^b]} \left| \frac{L_2(y)}{L_0(x)} \right| \le \limsup_{x \to 0} \left| \frac{L_2(x^b)}{L_0(x)} \right|.$$

Here, as $x \to 0$,

$$L_0(x) \to 2^{-2\nu+1} \frac{\Gamma\left(\frac{3}{2} - \nu\right)}{\Gamma\left(\nu + \frac{1}{2}\right)}.$$

while

$$L_2\left(x^b\right) \to a\left(\nu - \frac{1}{2}\right) \left\{ a\left(\frac{5}{2} - \nu\right)^{-1} - a\left(\frac{3}{2} - \nu\right)^{-1} \right\}.$$

Therefore also condition (A3) is satisfied.

6 An application

Let us consider the realised variation ratio (RVR) defined for a stochastic process X as

$$RVR_t^n := \frac{\frac{\pi}{2}V(X, 1, 1)_t^n}{V(X, 2, 0)_t^n}.$$
(6.1)

The RVR is of interest as a diagnostic tool concerning the nature of empirical processes.

In particular, it can be used to test the hypothesis that such a process is a Brownian semimartingale (with nontrivial local martingale component) against the possibility that it is of this type plus a jump process, see [BNSW06] and [Jac08c] (some related work is discussed in [Woe08]). If a jump component is present then the limit of RVR_t^n is smaller than 1.

However, in the course of the turbulence project, mentioned earlier, when calculating the RVR for an extensive high quality data set from atmospheric turbulence it turned out that the values of RVR were consistently higher than 1. The wish to understand this phenomenon has been a strong motivation for the theoretical developments described in the present paper. As a consequence of Theorem 3, we obtain the following probability limit result for the realised variation ratio of \mathcal{BSS} processes:

$$RVR_t^n - \psi\left(r_n(1)\right) \xrightarrow{ucp} 0 \tag{6.2}$$

where

$$\psi\left(\rho\right) = \sqrt{1 - \rho^2} + \rho \arcsin\rho, \qquad (6.3)$$

which equals $\frac{\pi}{2}$ times the mean $\mathbb{E}\{|UV|\}$ of two standard normal variables U and V with correlation ρ .

Remark 18 Notice that under the assumptions of Theorem 2 we have $r_n(1) \rightarrow \rho(1)$ and thus

$$RVR_t^n \xrightarrow{ucp} \psi(\rho(1)).$$

The latter can be used for parameter estimation. Under assumptions (A1)-(A3) we have that $\rho(1) = 2^{\alpha-1} - 1$. Consequently, the parameter α can be consistently estimated, because the function ψ is invertible on (0,1).

Moreover, we have that

$$\begin{split} \sqrt{n} \Big(RVR_t^n - \psi(r_n(1)) \Big) &= \sqrt{n} \Big(\frac{\frac{\pi}{2} V(Y, 1, 1)_t^n - \psi(r_n(1)) \int_0^t \sigma_s^2 \mathrm{d}s}{\int_0^t \sigma_s^2 \mathrm{d}s} \Big) \\ &- \sqrt{n} RVR_t^n \Big(\frac{V(Y, 2, 0)_t^n - \int_0^t \sigma_s^2 \mathrm{d}s}{\int_0^t \sigma_s^2 \mathrm{d}s} \Big), \end{split}$$

so, if the parameter $\alpha \in (0, 1)$, by applying Theorems 3 and 5, we obtain

$$\sqrt{n} \Big(RVR_t^n - \psi(r_n(1)) \Big) \xrightarrow{st} (\frac{\pi}{2}, -\psi(\rho(1))) \beta^{1/2} \frac{\int_0^t \sigma_s^2 \mathrm{d}B_s}{\int_0^t \sigma_s^2 \mathrm{d}s}, \tag{6.4}$$

where ψ is as above and the matrix β is given in Theorem 2. Specifically we find $\beta = (\beta_{ij})_{1 \le i,j \le 2}$ where

$$\beta_{11} = \lim_{n \to \infty} n \operatorname{var} \left(V_Q(1, 1)_1^n \right) ,$$

$$\beta_{22} = \lim_{n \to \infty} n \operatorname{var} \left(V_Q(2, 0)_1^n \right) ,$$

$$\beta_{12} = \lim_{n \to \infty} n \operatorname{cov} \left(V_Q(2, 0)_1^n, V_Q(1, 1)_1^n \right)$$

with Q as defined in Theorem 2. Thus, we obtain

$$\beta_{22} = \operatorname{var}(Q_1^2) + 2\sum_{k=1}^{\infty} \operatorname{cov}(Q_1^2, Q_{1+k}^2) = 2 + 4\sum_{k=1}^{\infty} \rho^2(k).$$

Similarly, we have that

$$\beta_{12} = \operatorname{cov}(Q_1^2, |Q_1||Q_2|) + 2\sum_{k=1}^{\infty} \operatorname{cov}(Q_1^2, |Q_{1+k}||Q_{2+k}|).$$

To compute the latter, we use the following formula:

$$E[|X_1^2 X_2 X_3|] = \frac{2}{\pi} \Big(\sqrt{1 - \rho_{23}^2} (1 + \rho_{12}^2 + \rho_{13}^2) + (\rho_{23} + 2\rho_{12}\rho_{13}) \operatorname{arcsin}(\rho_{23}) \Big) := h(\rho_{12}, \rho_{13}, \rho_{23}),$$

where X_1, X_2, X_3 are standard normal with $cov(X_i, X_j) = \rho_{ij}$, see [Nab52]. Consequently, we obtain the identity

$$\beta_{12} = \left(h(1,\rho(1),\rho(1)) - f(\rho(1))\right) + 2\sum_{k=1}^{\infty} \left(h(\rho(k),\rho(k+1),\rho(1)) - f(\rho(1))\right).$$

For the remaining term we deduce

$$\beta_{11} = \operatorname{var}(|Q_1||Q_2|) + 2\sum_{k=1}^{\infty} \operatorname{cov}(|Q_1||Q_2|, |Q_{1+k}||Q_{2+k}|).$$

However, there is no explicit formula available for the latter expression but it can be easily computed numerically.

7 Proofs

All positive constants (which do not depend on n) in the proof are denoted by C, although they might change from line to line.

Before we proceed with the proofs of the main results we review the basic concepts of the Wiener chaos expansion. Consider a complete probability space (Ω, \mathcal{F}, P) and a subspace \mathcal{H}_1 of $L^2(\Omega, \mathcal{F}, P)$ whose elements are zero-mean Gaussian random variables. Let $I\!H$ be a separable

Hilbert space with scalar product denoted by $\langle \cdot, \cdot \rangle_{I\!\!H}$ and norm $|| \cdot ||_{I\!\!H}$. We will assume that there is an isometry

$$W : I\!\!H \to \mathcal{H}_1$$
$$h \mapsto W(h)$$

in the sense that

$$E[W(h_1)W(h_2)] = \langle h_1, h_2 \rangle_{I\!H}.$$

It is easy to see that this map has to be linear.

For any $m \ge 2$, we denote by \mathcal{H}_m the *m*-th Wiener chaos, that is, the closed subspace of $L^2(\Omega, \mathcal{F}, P)$ generated by the random variables $H_m(X)$, where $X \in \mathcal{H}_1$, $E[X^2] = 1$, and H_m is the *m*-th Hermite polynomial, i.e. $H_0(x) = 1$ and $H_m(x) = (-1)^m e^{\frac{x^2}{2}} \frac{d^m}{dx^m} (e^{-\frac{x^2}{2}})$.

Suppose that $I\!H$ is infinite-dimensional and let $\{e_i, i \ge 1\}$ be an orthonormal basis of $I\!H$. Denote by Λ the set of all sequences $a = (a_1, a_2, ...), a_i \in \mathbb{N}$, such that all the terms, except a finite number of them, vanish. For $a \in \Lambda$ we set $a! = \prod_{i=1}^{\infty} a_i!$ and $|a| = \sum_{i=1}^{\infty} a_i$. For any multindex $a \in \Lambda$ we define

$$\Phi_a = \frac{1}{\sqrt{a!}} \prod_{i=1}^{\infty} H_{a_i}(W(e_i)).$$

The family of random variables $\{\Phi_a, a \in \Lambda\}$ is an orthonormal system. In fact

$$E\left[\prod_{i=1}^{\infty} H_{a_i}(W(e_i))\prod_{i=1}^{\infty} H_{b_i}(W(e_i))\right] = \delta_{ab}a!$$

where δ_{ab} denotes the Kronecker symbol. Moreover, $\{\Phi_a | a \in \Lambda, |a| = m\}$ is a complete orthonormal system in \mathcal{H}_m .

Let $a \in \Lambda$ with |a| = m. The mapping

$$\begin{split} I_m : I\!\!H^{\odot m} &\to \mathcal{H}_m \\ \widetilde{\otimes}_{i=1}^{\infty} e_i^{\otimes a_i} &\mapsto \Pi_{i=1}^{\infty} H_{a_i}(W(e_i)) \end{split}$$

between the symmetric tensor product $\mathbb{H}^{\odot m}$, equipped with the norm $\sqrt{m!} \|\cdot\|_{\mathbb{H}^{\otimes m}}$, and the *m*-th chaos \mathcal{H}_m is a linear isometry. Here $\widetilde{\otimes}$ denotes the symmetrization of the tensor product \otimes and I_0 is the identity in \mathbb{R} . For $h \in \mathbb{H}^{\otimes m}$ we set $I_m(h) := I_m(\tilde{h})$. For any $g \in \mathbb{H}^{\otimes m}$, $h \in \mathbb{H}^{\otimes n}$, $n, m \geq 0$, it holds that

$$E[I_m(g)I_n(h)] = \delta_{mn}m!\langle \tilde{g}, h \rangle_{I\!\!H^{\otimes m}}.$$

For any $h = h_1 \otimes \cdots \otimes h_m$ and $g = g_1 \otimes \cdots \otimes g_m \in \mathbb{H}^{\otimes m}$, we define the *p*-th contraction of *h* and *g*, denoted by $h \otimes_p g$, as the element of $\mathbb{H}^{\otimes 2(m-p)}$ given by

$$h \otimes_p g = \langle h_1, g_1 \rangle_{\mathbb{H}} \cdots \langle h_p, g_p \rangle_{\mathbb{H}} h_{p+1} \otimes \cdots \otimes h_m \otimes g_{p+1} \otimes \cdots \otimes g_m.$$

This definition can be extended by linearity to any element of $I\!\!H^{\otimes m}$.

Now, let \mathcal{G} be the σ -field generated by the random variables $\{W(h) | h \in \mathbb{H}\}$. Any square integrable random variable $F \in L^2(\Omega, \mathcal{G}, P)$ has a unique chaos decomposition

$$F = \sum_{m=0}^{\infty} I_m(h_m) \; ,$$

where $h_m \in I\!\!H^{\odot m}$ (see [Nu06] for more details).

Finally, we adapt the theory of Wiener chaos expansion to the set up of Section 3. Let \mathcal{G} be the σ -field generated by the random variables $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$ and \mathcal{H}_1 be the first Wiener chaos associated with $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$, i.e. the closed subspace of $L^2(\Omega, \mathcal{G}, P)$ generated by the random variables $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$. Notice that \mathcal{H}_1 can be seen as a separable Hilbert space with a scalar product induced by the covariance function of the process $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$. This means we can apply the above theory of Wiener chaos associated with the canonical Hilbert space $\mathbb{H} = \mathcal{H}_1$. Denote by \mathcal{H}_m the *m*th Wiener chaos associated with the triangular array $(X_{i,n})_{n\geq 1,1\leq i\leq [nt]}$ and by I_m the corresponding linear isometry between the symmetric tensor product $\mathcal{H}_1^{\odot m}$ (equipped with the norm $\sqrt{m!} \|\cdot\|_{\mathcal{H}^{\otimes m}}$) and the *m*th Wiener chaos.

7.1 Preliminary results

First of all, let us note that w.l.o.g. the volatility process σ can be assumed to be bounded on compact intervals because σ is càdlàg. This follows by a standard localization procedure presented in [BNGJPS06]. Furthermore, the process F_t , defined by (4.4), is continuous, because σ is càdlàg. Hence, F_t is locally bounded and can be assumed to be bounded on compact intervals w.l.o.g. by the same localization procedure.

Next we establish three lemmas.

Lemma 1 Under assumption (4.4) it holds that

$$E[|\Delta_i^n Y|^p] \le C_p \tau_n^p , \qquad i = 0, \dots, [nt]$$

$$(7.1)$$

for all p > 0.

Proof of Lemma 1: Recall that g' is non-increasing on (b, ∞) for some b > 0. Assume w.l.o.g. that b > 1. Observe the decomposition

$$\Delta_i^n Y = \int_{\frac{i-1}{n}}^{\frac{i}{n}} g(\frac{i}{n} - s)\sigma_s W(\mathrm{d}s) + \int_{-\infty}^{\frac{i-1}{n}} \left(g(\frac{i}{n} - s) - g(\frac{i-1}{n} - s)\right)\sigma_s W(\mathrm{d}s).$$

Since σ is bounded on compact intervals we deduce by Burkholder's inequality

$$E[|\Delta_{i}^{n}Y|^{p}] \leq C_{p}\left(\tau_{n}^{p} + E\left(\int_{0}^{\infty}\left(g(\frac{1}{n}+s) - g(s)\right)^{2}\sigma_{\frac{i-1}{n}-s}^{2}ds\right)^{p/2}\right)$$

We immediately obtain the estimates

$$\int_0^1 \left(g(\frac{1}{n} + s) - g(s) \right)^2 \sigma_{\frac{i-1}{n} - s}^2 ds \le C \tau_n^2 ,$$

O.E. Barndorff-Nielsen, J.M. Corcuera and M. Podolskij: Asymptotic results

$$\int_{1}^{b} \left(g(\frac{1}{n} + s) - g(s) \right)^{2} \sigma_{\frac{i-1}{n} - s}^{2} ds \le \frac{C}{n^{2}} ,$$

because g' is continuous on $(0, \infty)$ and σ is bounded on compact intervals. On the other hand, since g' is non-increasing on (b, ∞) , we get

$$\int_{b}^{\infty} \left(g(\frac{1}{n}+s) - g(s)\right)^{2} \sigma_{\frac{i-1}{n}-s}^{2} ds \leq \frac{F_{\frac{i-1}{n}}}{n^{2}}.$$

The boundedness of the process F implies (7.1).

Next, for any stochastic process f and any s > 0, we define the (possibly infinite) measure (recall that g(x) := 0 for $x \le 0$)

$$\pi_{f,s}^{n}(A) = \frac{E \int_{A} (g(x - \frac{1}{n}) - g(x))^{2} f_{s-x}^{2} \mathrm{d}x}{\int_{0}^{\infty} (g(x - \frac{1}{n}) - g(x))^{2} \mathrm{d}x} , \qquad A \in \mathcal{B}(\mathbb{R}).$$
(7.2)

We further define $\overline{\pi}_{f,s}^n(x) = \pi_{f,s}^n(\{y: y > x\}).$

Lemma 2 Under assumption (4.4) it holds that

$$\sup_{s \in [0,t]} \overline{\pi}^n_{\sigma,s}(\varepsilon) \le C \overline{\pi}^n(\varepsilon) \tag{7.3}$$

for any $\varepsilon > 0$, where $\overline{\pi}^n$ is given by (4.5).

Proof of Lemma 2: Recall again that g' is non-increasing on (b, ∞) for some b > 0, and assume w.l.o.g. that $b > \varepsilon$. Since the processes σ and F are bounded we deduce exactly as in the previous proof that

$$\begin{split} \int_{\varepsilon}^{\infty} (g(x-\frac{1}{n}) - g(x))^2 \sigma_{s-x}^2 \mathrm{d}x &= \int_{\varepsilon}^{b} (g(x-\frac{1}{n}) - g(x))^2 \sigma_{s-x}^2 \mathrm{d}x + \int_{b}^{\infty} (g(x-\frac{1}{n}) - g(x))^2 \sigma_{s-x}^2 \mathrm{d}x \\ &\leq C \Big(\int_{\varepsilon}^{\infty} (g(x-\frac{1}{n}) - g(x))^2 \mathrm{d}x + n^{-2} \Big). \end{split}$$

This completes the proof of Lemma 2.

Finally, we present the following technical Lemma.

Lemma 3 Under the assumption (CLT) there exists a number $l \ge 1$ and positive sequences $\varepsilon_n^{(j)} \to 0, j = 1, ..., l$, such that $0 < \varepsilon_n^{(1)} < \cdots < \varepsilon_n^{(l)}$ and

$$\varepsilon_n^{(1)} = o(n^{-\frac{1}{2\gamma(p\wedge 1)}}), \qquad \overline{\pi}^n(\varepsilon_n^{(l)}) = o(n^{-\frac{1}{p\wedge 1}})$$
(7.4)

$$(\varepsilon_n^{(j+1)})^{2\gamma} \overline{\pi}^n(\varepsilon_n^{(j)}) = o(n^{-\frac{1}{p\wedge 1}}) , \qquad j = 1, \dots, l-1 , \qquad (7.5)$$

where $p = \min_{1 \le i \le k, 1 \le j \le d} (p_i^j)$.

Proof of Lemma 3: Assume first that $p \ge 1$. Recall that $\gamma > 1/2$. Set $\varepsilon_n^{(j)} = n^{-\kappa_j}$, $j = 1, \ldots, l$, with $1 > \kappa_1 > \ldots > \kappa_l > 0$. The condition $\overline{\pi}^n(\varepsilon_n^{(j)}) = O(n^{\lambda(1-\kappa_j)})$ for some $\lambda < -1$, presented in (4.8), implies that conditions (7.4) and (7.5) are satisfied if we find $1 > \kappa_1 > \ldots > \kappa_l > 0$ such that

$$\begin{split} \kappa_1 &> \frac{1}{2\gamma} ,\\ \kappa_l &< 1 + \frac{1}{\lambda} ,\\ (1+\lambda) - \kappa_j \lambda - 2\kappa_{j+1}\gamma < 0 , \qquad 1 \le j \le l-1. \end{split}$$

From the first and the last inequality we deduce by induction that

$$\frac{1}{2\gamma} < \kappa_1 < \frac{1+\lambda}{\lambda} \sum_{i=0}^{l-1} \left(-\frac{2\gamma}{\lambda}\right)^i + \left(-\frac{2\gamma}{\lambda}\right)^l \kappa_l$$

must hold.

When $2\gamma \geq -\lambda$ the term on the right-hand side converges to ∞ as $l \to \infty$. In that case it

is easy to find constants $1 > \kappa_1 > \ldots > \kappa_l > 0$ such that (7.4) and (7.5) are satisfied. When $2\gamma < -\lambda$ the limit of $\frac{1+\lambda}{\lambda} \sum_{i=0}^{l-1} \left(-\frac{2\gamma}{\lambda}\right)^i$ is $\frac{1+\lambda}{\lambda+2\gamma}$ (as $l \to \infty$) and the restriction on κ_1 becomes

$$\frac{1}{2\gamma} < \kappa_1 < \frac{1+\lambda}{\lambda+2\gamma}.$$

Notice that $\frac{1}{2\gamma} < \frac{1+\lambda}{\lambda+2\gamma}$ because $\gamma > 1/2$. The existence of the positive powers κ_j , $j = 2, \ldots, l$ that satisfy the original inequality follows by an induction argument.

Assume now that p < 1. Recall that γ must satisfy

$$\gamma > \frac{1}{2p}$$

and $\lambda < -\frac{1}{p}$. Again the conditions (7.4) and (7.5) are satisfied if we find $1 > \kappa_1 > \ldots > \kappa_l > 0$ such that

$$\begin{split} \kappa_1 &> \frac{1}{2\gamma p} \ , \\ \kappa_l &< \frac{1+\lambda p}{\lambda p} \ , \\ (\frac{1}{p}+\lambda) - \kappa_j \lambda - 2\kappa_{j+1}\gamma < 0, \qquad 1 \le j \le l-1. \end{split}$$

Notice that the second inequality has solutions because $\lambda < -\frac{1}{p}$. Moreover, we deduce as above that the inequality

$$\frac{1}{2\gamma p} < \kappa_1 < \frac{\frac{1}{p} + \lambda}{\lambda} \sum_{i=0}^{l-1} \left(-\frac{2\gamma}{\lambda} \right)^i + \left(-\frac{2\gamma}{\lambda} \right)^l \kappa_l$$

must hold. Again the more complicated case is $2\gamma < -\lambda$. By letting $l \to \infty$ the restriction on κ_1 becomes

$$\frac{1}{2\gamma p} < \kappa_1 < \frac{\frac{1}{p} + \lambda}{\lambda + 2\gamma}$$

Note that $\frac{1}{2\gamma p} < \frac{\frac{1}{p} + \lambda}{\lambda + 2\gamma}$ because $\gamma > \frac{1}{2p}$. As before the existence of the positive powers κ_j , $j = 2, \ldots, l$ that satisfy the original inequality follows by an induction argument. \Box

7.2 Some notation

Recall that the covariance matrix of $(X_{i,n}, \ldots, X_{i+l,n})$ is invertible for any $l \ge 1$ and $n \ge 1$. 1. Let $X_i^n(1), \ldots, X_i^n(k)$ be an i.i.d. N(0,1) sequence that spans the same linear space as $X_{i,n}, \ldots, X_{i+k-1,n}$ (such a sequence can be constructed by the Gram-Schmidt method). Thus, it has the representation

$$X_i^n(j) = \sum_{l=1}^k a_{lj}^n X_{i+l-1,n} , \qquad j = 1, \dots, k , \qquad (7.6)$$

for some real numbers a_{lj}^n . Note that

$$|a_{lj}^n| \le C \; ,$$

for all l, j, n, because $E[X_{i,n}^2] = 1$ for all i, n.

For any $1 \leq j \leq d$, we obtain the Wiener chaos representation

$$V(p_1^j, \dots, p_k^j)_t^n - \rho_{p_1^j, \dots, p_k^j}^{(n)} t = \sum_{m=2}^{\infty} I_m \left(\frac{1}{n} \sum_{i=1}^{[nt]} f_{m,j}^n(i)\right) + O_p(n^{-1}) , \qquad (7.7)$$

where the $f_{m,j}^n(i) \in I\!\!H^{\odot m}$ are given by

$$f_{m,j}^{n}(i) = \sum_{k_l \in \{1,\dots,k\}} c_{k_1,\dots,k_m}^{n}(j) X_i^{n}(k_1) \otimes \dots \otimes X_i^{n}(k_m)$$
(7.8)

for some coefficients $c_{k_1,\ldots,k_m}^n(j)$. We set

$$c_m^n(j) = ||f_{m,j}^n(i)||_{H^{\otimes m}}^2 = \sum_{k_l \in \{1,\dots,k\}} |c_{k_1,\dots,k_m}^n(j)|^2.$$
(7.9)

Note that

$$\operatorname{var}\left(|X_{i,n}|^{p_1^j} \cdots |X_{i+k-1,n}|^{p_k^j}\right) = \sum_{m=2}^{\infty} m! c_m^n(j) < C$$
(7.10)

for all n, j, because $E[X_{i,n}^2] = 1$ for all i, n. Finally, when $f_{m,j}^n(i)$, $c_{k_1,\ldots,k_m}^n(j)$ and $c_m^n(j)$ correspond to some particular choice of powers p_1, \ldots, p_k we use the notation $f_m^n(i)$, c_{k_1,\ldots,k_m}^n and c_m^n .

Now assume that the assumptions (3.8) and (3.9) of Theorem 2 hold. Since a_{lj}^n in (7.6) is a continuous function of $r(1), \ldots, r_n(k-1)$ and the Gaussian process Q is non-degenerate, we have that

$$a_{lj}^n \to a_{lj}$$

and the sequence $Q_i(1), \ldots, Q_i(k)$ given by

$$Q_i(j) = \sum_{l=1}^k a_{lj} Q_{i+l-1} , \qquad j = 1, \dots, k , \qquad (7.11)$$

is an i.i.d. N(0,1) sequence. Now, let us associate $f_{m,j}(i)$, $c_{k_1,\ldots,k_m}(j)$ and $c_m(j)$ with the functional $V_Q(p_1^j,\ldots,p_k^j)_t^n - \rho_{p_1^j,\ldots,p_k^j}t$, where

$$\rho_{p_1^j,\dots,p_k^j} = E\Big[|Q_1|^{p_1^j}\cdots|Q_k|^{p_k^j}\Big],$$

by (7.7), (7.8) and (7.9). By a repeated application of the multiplication formula (see [Nu06]), we know that $c_{k_1,\ldots,k_m}^n(j)$ is a continuous function of $r_n(1),\ldots,r_n(k-1)$. Since $r_n(j) \to \rho(j)$ we obtain

$$c_{k_1,\dots,k_m}^n(j) \to c_{k_1,\dots,k_m}(j) , \qquad c_m^n(j) \to c_m(j) ,$$
 (7.12)

$$\langle f_{m,j_1}^n(i), f_{m,j_2}^n(i+l) \rangle_{\mathbb{H}^{\otimes m}} \to \langle f_{m,j_1}(i), f_{m,j_2}(i+l) \rangle_{\mathbb{H}^{\otimes m}},$$
(7.13)

$$\operatorname{cov}\left(|X_{i,n}|^{p_{1}^{j_{1}}}\cdots|X_{i+k-1,n}|^{p_{k}^{j_{1}}},|X_{i,n}|^{p_{1}^{j_{2}}}\cdots|X_{i+k-1,n}|^{p_{k}^{j_{2}}}\right) = \sum_{m=2}^{\infty} m! \langle f_{m,j_{1}}^{n}(1), f_{m,j_{2}}^{n}(1) \rangle_{H^{\otimes m}}$$

$$\to \operatorname{cov}\left(|Q_{i}|^{p_{1}^{j_{1}}}\cdots|Q_{i+k-1}|^{p_{k}^{j_{1}}},|X_{i,n}|^{p_{1}^{j_{2}}}\cdots|Q_{i+k-1}|^{p_{k}^{j_{2}}}\right)$$

$$= \sum_{m=2}^{\infty} m! \langle f_{m,j_{1}}(1), f_{m,j_{2}}(1) \rangle_{H^{\otimes m}}.$$
(7.14)

7.3 Proof of Theorem 1 and 3

Proof of Theorem 1: Since $V(p_1, \ldots, p_k)_t^n$ is increasing in t and the process $\rho_{p_1, \ldots, p_k}^{(n)} t$ is continuous in t, it is sufficient to prove $V(p_1, \ldots, p_k)_t^n - \rho_{p_1, \ldots, p_k}^{(n)} t \xrightarrow{P} 0$ for a fixed t > 0.

Note that

$$|\langle f_m^n(1), f_m^n(1+l) \rangle_{\mathbb{H}^{\otimes m}}| \le c_m^n, \quad |\langle f_m^n(1), f_m^n(1+l) \rangle_{\mathbb{H}^{\otimes m}}| \le c_m^n C^m(|r_n(l)|^m + \dots + |r_n(l-k+1)|^m)$$
(7.15)

Now, due to assumption (3.3), $r(j) \to 0$ as $j \to \infty$. Thus, there exists a H such that $|Cr^{1/2}(j-k+1)| < 1$ for $j \ge H$ (for any fixed C). By (7.15) we have (for any $m \ge 2$)

$$\sum_{l=1}^{n-1} |\langle f_m^n(1), f_m^n(1+l) \rangle_{H^{\otimes m}}| \le C \Big(H c_m^n + \sum_{l=H}^{n-1} |\langle f_m^n(1), f_m^n(1+l) \rangle_{H^{\otimes m}}| \Big)$$

$$\le C c_m^n \Big(H + \sum_{l=H}^{n-1} (C |r_n(l)|)^2 \Big) \le C c_m^n \sum_{l=1}^{n-1} r(l).$$
(7.16)

Hence,

$$\operatorname{var}(V(p_1,\ldots,p_k)_t^n) \le \frac{C}{n} \sum_{m=2}^{\infty} m! c_m^n \Big(1 + \sum_{l=1}^{n-1} r(l) \Big).$$

The latter converges to 0 due to (7.10) and assumption (3.3).

Proof of Theorem 3: First of all, recall that

$$E[|\Delta_i^n Y|^q] \le C\tau_n^q , \qquad E[|\Delta_i^n G|^q] \le C\tau_n^q , \qquad (7.17)$$

for any $q \ge 0$, due to Lemma 1.

In the following we will prove Theorem 3 only for k = 1, $p_1 = p$. The general case can be obtained in a similar manner by (7.17) and an application of the Hölder inequality.

Since $V(Y, p)_t^n$ is increasing in t and the limit process is continuous in t, it suffices to prove the pointwise convergence $V(Y, p)_t^n \xrightarrow{P} \mu_p \int_0^t |\sigma_s|^p ds$. For any $l \leq n$, we have

$$V(Y,p)_t^n - \mu_p \int_0^t |\sigma_s|^p ds = \frac{1}{n\tau_n^p} \sum_{i=1}^{[nt]} \left(|\Delta_i^n Y|^p - |\sigma_{\frac{i-1}{n}} \Delta_i^n G|^p \right) + R_t^{n,l} ,$$

where

$$\begin{split} R_t^{n,l} &= \frac{1}{n\tau_n^p} \Big(\sum_{i=1}^{[nt]} |\sigma_{\frac{i-1}{n}} \Delta_i^n G|^p - \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \sum_{i \in I_l(j)} |\Delta_i^n G|^p \Big) \\ &+ \frac{1}{n\tau_n^p} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \sum_{i \in I_l(j)} |\Delta_i^n G|^p - \mu_p l^{-1} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \\ &+ \mu_p \Big(l^{-1} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p - \int_0^t |\sigma_s|^p \, ds \Big) \;, \end{split}$$

and

$$I_l(j) = \left\{ i \mid \frac{i}{n} \in \left(\frac{j-1}{l}, \frac{j}{l}\right] \right\}, \qquad j \ge 1.$$

The assumption **(LLN)** implies that $V(G,p)_t^n \xrightarrow{ucp} \mu_p t$. Since σ is càdlàg and bounded on compact intervals, we deduce that

$$\lim_{l \to \infty} \lim_{n \to \infty} P(|R_t^{n,l}| > \epsilon) = 0 ,$$

for any $\epsilon > 0$. Hence, we are left to prove that

$$\frac{1}{n\tau_n^p}\sum_{i=1}^{[nt]} \left(|\Delta_i^n Y|^p - |\sigma_{\frac{i-1}{n}} \Delta_i^n G|^p \right) \stackrel{P}{\longrightarrow} 0.$$

By applying the inequality $||x|^p - |y|^p| \le p|x - y|(|x|^{p-1} + |y|^{p-1})$ for p > 1 and $||x|^p - |y|^p| \le |x - y|^p$ for $p \le 1$, (7.17) and the Cauchy-Schwarz inequality we can conclude that the above convergence follows from

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|\Delta_i^n Y - \sigma_{\frac{i-1}{n}} \Delta_i^n G|^2] \to 0.$$
(7.18)

Observe the decomposition

$$\Delta_i^n Y - \sigma_{\frac{i-1}{n}} \Delta_i^n G = A_i^n + B_i^{n,\varepsilon} + C_i^{n,\varepsilon} ,$$

where

$$A_i^n = \int_{\frac{i-1}{n}}^{\frac{i}{n}} g\left(\frac{i}{n} - s\right) (\sigma_s - \sigma_{\frac{i-1}{n}}) W(\mathrm{d}s)$$

$$B_i^{n,\varepsilon} = \int_{\frac{i-1}{n}-\varepsilon}^{\frac{i-1}{n}} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right) \sigma_s W(\mathrm{d}s) - \sigma_{\frac{i-1}{n}} \int_{\frac{i-1}{n}-\varepsilon}^{\frac{i-1}{n}} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right) W(\mathrm{d}s)$$

$$C_i^{n,\varepsilon} = \int_{-\infty}^{\frac{i-1}{n}-\varepsilon} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right) \sigma_s W(\mathrm{d}s) - \sigma_{\frac{i-1}{n}} \int_{-\infty}^{\frac{i-1}{n}-\varepsilon} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right) W(\mathrm{d}s)$$

By Lemma 2 and the boundedness of σ on compact intervals we deduce

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|C_i^{n,\varepsilon}|^2] \le C\overline{\pi}^n(\varepsilon).$$
(7.19)

and by (4.6) we obtain that

$$\lim_{n \to \infty} \frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|C_i^{n,\varepsilon}|^2] = 0.$$

Next, we get

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|A_i^n|^2] \le \frac{C}{n\tau_n^2} E\Big[\sum_{i=1}^{[nt]} \int_{\frac{i-1}{n}}^{\frac{i}{n}} g^2\Big(\frac{i}{n} - s\Big)(\sigma_s - \sigma_{\frac{i-1}{n}})^2 ds\Big].$$
(7.20)

Set $v(s,\eta) = \sup\{|\sigma_s - \sigma_r|^2 | s, r \in [-t,t], |r-s| \le \eta\}$. Then we obtain

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|A_i^n|^2] \le \frac{1}{n} \sum_{i=1}^{[nt]} E\left[v(\frac{i-1}{n}, n^{-1})\right].$$
(7.21)

Moreover, for any $\kappa > 0$, since σ is cadlag, there exists n big enough such that

$$v(\frac{i-1}{n}, n^{-1}) \le \kappa + \left(\Delta \sigma_{\frac{i-1}{n}}\right)^2 \mathbf{1}_{\left\{\left(\Delta \sigma_{\frac{i-1}{n}}\right)^2 \ge \kappa\right\}},$$

 \mathbf{SO}

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|A_i^n|^2] \leq \kappa + \frac{1}{n} \sum_{i=1}^{[nt]} E\Big[\left(\Delta\sigma_{\frac{i-1}{n}}\right)^2 \mathbf{1}_{\left\{\left(\Delta\sigma_{\frac{i-1}{n}}\right)^2 \ge \kappa\right\}}\Big)\Big] \\
\leq \kappa + E\Big[\frac{1}{n} \sum_{-t \le s \le t} (\Delta\sigma_s)^2 \mathbf{1}_{\left\{(\Delta\sigma_s)^2 \ge \kappa\right\}}\Big)\Big],$$

then

$$\lim_{n \to \infty} \frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|A_i^n|^2] \le \kappa$$

and the convergence to zero follows letting κ tend to zero.

Finally, observe the decomposition $B_i^{n,\varepsilon}=B_i^{n,\varepsilon}(1)+B_i^{n,\varepsilon}(2)$ with

$$B_{i}^{n,\varepsilon}(1) = \int_{\frac{i-1}{n}-\varepsilon}^{\frac{i-1}{n}} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right)(\sigma_{s}-\sigma_{\frac{i-1}{n}-\varepsilon})W(\mathrm{d}s)$$
$$B_{i}^{n,\varepsilon}(2) = \left(\sigma_{\frac{i-1}{n}-\varepsilon} - \sigma_{\frac{i-1}{n}}\right)\int_{\frac{i-1}{n}-\varepsilon}^{\frac{i-1}{n}} \left(g\left(\frac{i}{n}-s\right) - g\left(\frac{i-1}{n}-s\right)\right)W(\mathrm{d}s)..$$

We obtain the inequalities

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|B_i^{n,\varepsilon}(1)|^2] \le \frac{1}{n} \sum_{i=1}^{[nt]} E\left[v(\frac{i-1}{n},\varepsilon)\right]$$

$$\frac{1}{n\tau_n^2} \sum_{i=1}^{[nt]} E[|B_i^{n,\varepsilon}(2)|^2] \le \frac{1}{n} \sum_{i=1}^{[nt]} E\left[v(\frac{i-1}{n},\varepsilon)^2\right]^{1/2}.$$
(7.22)

By using the same arguments as above we have that both terms converge to zero and we obtain (7.18), which completes the proof of Theorem 3.

7.4 Proof of Theorem 2 and 4

Proof of Theorem 2: We first show the weak convergence of finite dimensional distributions and then prove the tightness of the sequence $\sqrt{n} \left(V(p_1^j, \ldots, p_k^j)_t^n - \rho_{p_1^j, \ldots, p_k^j}^{(n)} t \right)_{1 \le j \le d}$.

Step 1: Define the vector $Z_n(j) = (Z_n^1(j), \dots, Z_n^e(j))^T$, $1 \le j \le d$, by

$$Z_n^l(j) = \frac{1}{\sqrt{n}} \sum_{i=[nc_l]+1}^{[nb_l]} \left(|X_{i,n}|^{p_1^j} \cdots |X_{i+k-1,n}|^{p_k^j} - \rho_{p_1^j,\dots,p_k^j}^{(n)} \right),$$
(7.23)

where $(c_l, b_l]$, l = 1, ..., e, are disjoint intervals contained in [0, T]. Set $Z_n^l = (Z_n^l(1), ..., Z_n^l(d))$, l = 1, ..., e. Clearly, it suffices to prove that

$$\left(Z_n^l\right)_{1\leq l\leq e} \xrightarrow{\mathcal{D}} \left(\beta^{1/2}(B_{b_l}-B_{c_l})\right)_{1\leq l\leq e},$$

where the matrix β is given in Theorem 2. By (7.7) we have the representation

$$Z_n^l(j) = \sum_{m=2}^{\infty} I_k \Big(\frac{1}{\sqrt{n}} \sum_{i=[nc_l]+1}^{[nb_l]} f_{m,j}^n(i) \Big).$$

Set $F_{m,l}^n(j) = \frac{1}{\sqrt{n}} \sum_{i=[nc_l]+1}^{[nb_l]} f_{m,j}^n(i)$. By Theorem 2 in [BNCPW09] we obtain the weak convergence of finite dimensional distributions when we show that

(i) For any $1 \le l \le e, 1 \le j \le d$ we have

$$\lim_{N \to \infty} \limsup_{n \to \infty} \sum_{m=N+1}^{\infty} m! ||F_{m,l}^n(j)||_{H^{\otimes k}}^2 = 0.$$

(ii) For any $m \ge 2, 1 \le l \le e$ and $1 \le j_1, j_2 \le d$, we have constants $C_{k,l}$ such that

 $\lim_{n \to \infty} m! \langle F_{m,l}^n(j_1), F_{m,l}^n(j_2) \rangle_{H^{\otimes m}} = C_{m,l}(j_1, j_2) ,$

and $\sum_{m=2}^{\infty} C_{m,l}(j_1, j_2) = \beta_{j_1, j_2}(b_l - c_l).$

(iii) For any $1 \le l_1 \ne l_2 \le e$ and $1 \le j_1, j_2 \le d$, we have

$$\lim_{n \to \infty} \langle F_{m,l_1}^n(j_1), F_{m,l_2}^n(j_2) \rangle_{\mathbb{H}^{\otimes m}} = 0.$$

(iv) For any $m \ge 2, 1 \le l \le e, 1 \le j \le d$ and $p = 1, \dots, m - 1$

$$\lim_{n \to \infty} ||F_{m,l}^n(j) \otimes_p F_{m,l}^n(j)||_{H^{\otimes 2(m-p)}}^2 = 0.$$

Note that it is sufficient to prove (i), (ii) and (iv) for l = 1, $b_l = 1$ and $a_l = 0$. In this case we use the notation $F_m^n(j) = F_{m,1}^n(j)$.

(i) and (ii): As in (7.16) we have

$$m! \langle F_{m}^{n}(j_{1}), F_{m}^{n}(j_{2}) \rangle_{H^{\otimes m}} = m! \Big(\langle f_{m,j_{1}}^{n}(1), f_{m,j_{2}}^{n}(1) \rangle_{H^{\otimes m}} + \frac{2}{n} \sum_{l=1}^{n-1} (n-l) \langle f_{m,j_{1}}^{n}(1), f_{m,j_{2}}^{n}(1+l) \rangle_{H^{\otimes m}} \Big)$$

$$\leq Cm! \langle f_{m,j_{1}}^{n}(1), f_{m,j_{2}}^{n}(1) \rangle_{H^{\otimes m}} \Big(1 + \sum_{l=1}^{n-1} r(l) \Big)$$
(7.24)

Since $\sum_{l=1}^{\infty} r(l) < \infty$, we obtain by (7.12)-(7.14) and the dominated convergence theorem

$$\lim_{n \to \infty} m! \langle F_{m,l}^n(j_1), F_{m,l}^n(j_2) \rangle_{\mathbb{H}^{\otimes m}} = C_m(j_1, j_2)$$
$$= m! \Big(\langle f_{m,j_1}(1), f_{m,j_2}(1) \rangle_{\mathbb{H}^{\otimes m}} + 2 \sum_{l=1}^{\infty} \langle f_{m,j_1}^n(1), f_{m,j_2}^n(1+l) \rangle_{\mathbb{H}^{\otimes m}} \Big) ,$$

and $\sum_{m=2}^{\infty} C_m(j_1, j_2) = \beta_{j_1, j_2}$ (notice that β_{j_1, j_2} is finite due to the dominated convergence theorem). Hence, we deduce (ii). On the other hand, we have

$$\limsup_{n \to \infty} \sum_{m=N+1}^{\infty} m! \langle f_{m,j_1}^n(1), f_{m,j_2}^n(1) \rangle_{H^{\otimes m}} = \sum_{m=N+1}^{\infty} m! \langle f_{m,j_1}(1), f_{m,j_2}(1) \rangle_{H^{\otimes m}} < \infty.$$

Thus, we obtain (i) by (7.24).

(iii): W.l.o.g. consider the case $j = j_1 = j_2$. For any $l_1 < l_2$, as in (7.16), we have

$$|\langle F_{m,l_1}^n(j), F_{m,l_2}^n(j) \rangle_{\mathbb{H}^{\otimes m}}| \le \frac{C}{n} \sum_{h=[nc_{l_1}]+1}^{[nb_{l_1}]} \sum_{i=[nc_{l_2}]+1}^{[nb_{l_2}]} |r_n^m(i-h)|.$$

Assume w.l.o.g. that $c_{l_1} = 0$, $b_{l_1} = c_{l_2} = 1$ and $b_{l_2} = 2$ (the case $b_{l_1} < c_{l_2}$ is much easier). Then, by condition (3.9), we obtain the approximation (as in (7.24))

$$|\langle F_{m,l_1}^n(j), F_{m,l_2}^n(j) \rangle_{\mathbb{H}^{\otimes m}}| \le C \Big(\frac{1}{n} \sum_{j=1}^n jr(j) + \sum_{j=1}^{n-1} r(n+j) \Big) \to 0 ,$$

since $\sum_{j=1}^{\infty} r(j) < \infty$.

(iv): A straightforward computation shows that

$$||F_m^n(j)\otimes_p F_m^n(j)||_{H^{\otimes 2(m-p)}}^2 = \frac{C_{m,j}}{n^2} \sum_{i_1,i_2,i_3,i_4=1}^{n-1} r_n^p(|i_1-i_2|)r_n^p(|i_4-i_3|)r_n^{m-p}(|i_1-i_4|)r_n^{m-p}(|i_2-j_3|).$$

for some constant $C_{m,j}$. The latter is smaller than

$$\frac{C}{n}\sum_{i,h,l=1}^{n-1}|r_n^p(i)||r_n^p(l)||r_n^{m-p}(|i-h|)||r_n^{m-p}(|l-h|)| = \frac{C}{n}\sum_{h=1}^{n-1}\Big(\sum_{i=1}^{n-1}|r_n^p(i)||r_n^{m-p}(|i-h|)|\Big)^2.$$

Now, for any $0 < \varepsilon < 1$, we obtain by the Hölder inequality

$$n^{-1} \sum_{0 \le h \le n-1} \left(\sum_{0 \le i \le n-1} |r_n^p(i)| |r_n^{m-p}(|i-h|)| \right)^2 \le n^{-1} \sum_{0 \le h \le [n\varepsilon]} \left(\sum_{0 \le i \le n-1} |r_n^p(i)| |r_n^{m-p}(|i-h|)| \right)^2 + 2n^{-1} \sum_{h=[n\varepsilon]}^{n-1} \left(\sum_{i=0}^{n-1} |r_n^p(i)| |r_n^{m-p}(|i-h|)| \right)^2 + 2n^{-1} \sum_{h=[n\varepsilon]}^{n-1} \left(\sum_{h=[n\varepsilon/2]}^{n-1} |r_n^p(i)| |r_n^{m-p}(|i-h|)| \right)^2 \le C \left(\varepsilon \left(\sum_{0 \le i \le n-1} |r_n^m(i)| \right)^2 + \left(\sum_{0 \le i \le n-1} |r_n^m(i)| \right)^{2p/m} \left(\sum_{[n\varepsilon/2] < h \le n-1} |r_n^m(h)| \right)^{2(m-p)/m} \right)$$

The latter is smaller (again by (3.9)) than

$$C\left(\varepsilon\left(\sum_{0\leq i\leq n-1}r(i)\right)^2 + \left(\sum_{0\leq i\leq n-1}r(i)\right)^{2p/m}\left(\sum_{[n\varepsilon/2]< h\leq n-1}r(h)\right)^{2(m-p)/m}\right)$$

that converges to $C\varepsilon \left(\sum_{i=0}^{\infty} r(i)\right)^2$ as $n \to \infty$. Thus, we obtain (iv) by letting $\varepsilon \to 0$.

Step 2: Clearly, it suffices to consider the case $d = 1, p_l^1 = p_l$. Set

$$\sqrt{n}\Big(V(p_1,\ldots,p_k)_t^n - \rho_{p_1,\ldots,p_k}^{(n)}t\Big) = \sum_{m=2}^{\infty} I_m\Big(\frac{1}{\sqrt{n}}\sum_{i=1}^{[nt]} f_m^n(i)\Big) + O_p(n^{-1/2}) =: Z_t^n + O(n^{-1/2})$$

(where the approximation holds locally uniformly in t) and

$$Z_t^{n,N} = \sum_{m=2}^N I_m \left(\frac{1}{\sqrt{n}} \sum_{i=1}^{[nt]} f_m^n(i) \right).$$

In Step 1 we have proved that conditions (i)-(iii) of Theorem 2 in [BNCPW09] are satisfied. Then by (2.3) of Theorem 2 in [BNCPW09] and the Cauchy-Schwarz inequality we obtain the approximation

$$\begin{split} &P\Big(|Z_t^{n,N} - Z_{t_1}^{n,N}| \ge \lambda, |Z_{t_2}^{n,N} - Z_t^{n,N}| \ge \lambda\Big) \le \frac{E^{1/2}[|Z_t^{n,N} - Z_{t_1}^{n,N}|^4]E^{1/2}[|Z_{t_2}^{n,N} - Z_t^{n,N}|^4]}{\lambda^4} \\ &\le \quad C\frac{\beta^2([nt] - [nt_1])([nt_2] - [nt])}{\lambda^4} \le C\frac{\beta^2(t_2 - t_1)^2}{\lambda^4} \end{split}$$

for any $t_1 \leq t \leq t_2$ and $\lambda > 0$. On the other hand (7.14) and (7.24) imply that

$$\lim_{N \to \infty} E[|Z_t^n - Z_t^{n,N}|^2] = 0$$

for any n and any t. Using this we conclude that

$$P(|Z_t^n - Z_{t_1}^n| \ge \lambda, |Z_{t_2}^n - Z_t^n| \ge \lambda) \le C \frac{\beta^2 (t_2 - t_1)^2}{\lambda^4}$$

for any $t_1 \leq t \leq t_2$ and $\lambda > 0$, from which we deduce the tightness of the sequence Z_t^n by Theorem 15.6 in [Bil68]. This completes the proof of Theorem 2.

Proof of Theorem 5: We only consider the case d = 1, k = 1, $p_1 = p$ (the general result is obtained by analogous arguments). We use the decomposition from the proof of Theorem 3:

$$\sqrt{n} \Big(V(Y,p)_{t}^{n} - \mu_{p} \int_{0}^{t} |\sigma_{s}|^{p} ds \Big) = \sqrt{n} \Big(\frac{1}{n\tau_{n}^{p}} \sum_{j=1}^{[lt]} |\sigma_{j-1}|^{p} \sum_{i \in I_{l}(j)} |\Delta_{i}^{n}G|^{p} - \mu_{p}l^{-1} \sum_{j=1}^{[lt]} |\sigma_{j-1}|^{p} \Big) \\
+ \frac{1}{\sqrt{n}\tau_{n}^{p}} \sum_{i=1}^{[nt]} \Big(|\Delta_{i}^{n}Y|^{p} - |\sigma_{j-1}\Delta_{i}^{n}G|^{p} \Big) + \overline{R}_{t}^{n,l} ,$$
(7.25)

for any $l \leq n$, with

$$\overline{R}_{t}^{n,l} = \frac{1}{\sqrt{n}\tau_{n}^{p}} \Big(\sum_{i=1}^{[nt]} |\sigma_{\frac{i-1}{n}} \Delta_{i}^{n}G|^{p} - \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^{p} \sum_{i \in I_{l}(j)} |\Delta_{i}^{n}G|^{p} \Big)$$
$$+ \sqrt{n}\mu_{p} \Big(l^{-1} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^{p} - \int_{0}^{t} |\sigma_{s}|^{p} ds \Big).$$

Observe that under the assumption (CLT) we obtain the weak convergence

$$\sqrt{n}\Big(V(G,p)_t^n - \mu_p t\Big) \Rightarrow \sqrt{\beta}B_t$$

(see Theorem 2). Since $E[G_t(V(G, p)_t^n - \mu_p t)] = 0$ for any t > 0, because G has a symmetric distribution, we deduce (by Theorem 5 in [BNCP09]) that

$$\left(G_t, \sqrt{n}\left(V(G, p)_t^n - \mu_p t\right)\right) \Rightarrow \left(G_t, \sqrt{\beta}B_t\right).$$

It follows that

$$\sqrt{n} \left(\frac{1}{n\tau_n^p} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \sum_{i \in I_l(j)} |\Delta_i^n G|^p - \mu_p l^{-1} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \right) \xrightarrow{\mathcal{G}-st} \sqrt{\beta} \sum_{j=1}^{[lt]} |\sigma_{\frac{j-1}{l}}|^p \Delta_j^l B$$

for any fixed l, because σ is assumed to be \mathcal{G} -measurable. On the other hand, we have

$$\sqrt{\beta} \sum_{j=1}^{[lt]} |\sigma_{j-1}|^p \Delta_j^l B \xrightarrow{P} \sqrt{\beta} \int_0^t |\sigma_s|^p \mathrm{d}B_s$$

as $l \to \infty$.

Now we need to prove that the other summands in the decomposition (7.25) are negligible. The negligibility of the term $\overline{R}_t^{n,l}$ is shown as in the proof of Theorem 7 in [BNCP09] but by using condition (4.7) instead of Hölder continuity of index γ . So we are left to prove that

$$\frac{1}{\sqrt{n}\tau_n^p}\sum_{i=1}^{[nt]} \left(|\Delta_i^n Y|^p - |\sigma_{\frac{i-1}{n}}\Delta_i^n G|^p \right) \xrightarrow{P} 0.$$

By applying, for $p \ge 1$, the inequality $||x|^p - |y|^p| \le p|x - y|(|x|^{p-1} + |y|^{p-1})$, (7.17) and the Cauchy-Schwarz inequality, and, for $p \le 1$, $||x|^p - |y|^p| \le |x - y|^p$ and the Jensen inequality, we have

$$\frac{1}{\sqrt{n}\tau_n^p} \sum_{i=1}^{[nt]} E \left| |\Delta_i^n Y|^p - |\sigma_{\frac{i-1}{n}} \Delta_i^n G|^p \right| \le \frac{1}{\sqrt{n}\tau_n^{p\wedge 1}} \sum_{i=1}^{[nt]} \left(E |\Delta_i^n Y - \sigma_{\frac{i-1}{n}} \Delta_i^n G|^2 \right)^{\frac{p\wedge 1}{2}}.$$

Now we use a similar decomposition as presented in the proof of Theorem 3:

$$\Delta_i^n Y - \sigma_{\frac{i-1}{n}} \Delta_i^n G = A_i^n + B_i^{n,\varepsilon_n^{(1)}} + \sum_{j=1}^l C_i^{n,\varepsilon_n^{(j)},\varepsilon_n^{(j+1)}}$$

where A_i^n , $B_i^{n,\varepsilon_n^{(1)}}$ are defined as above, $0 < \varepsilon_n^{(1)} < \cdots < \varepsilon_n^{(l)} < \varepsilon_n^{(l+1)} = \infty$ and

$$C_{i}^{n,\varepsilon_{n}^{(j)},\varepsilon_{n}^{(j+1)}} = \int_{\frac{i-1}{n}-\varepsilon_{n}^{(j+1)}}^{\frac{i-1}{n}-\varepsilon_{n}^{(j)}} \left(g\left(\frac{i}{n}-s\right)-g\left(\frac{i-1}{n}-s\right)\right)\sigma_{s}W(\mathrm{d}s)$$
$$-\sigma_{\frac{i-1}{n}}\int_{\frac{i-1}{n}-\varepsilon_{n}^{(j+1)}}^{\frac{i-1}{n}-\varepsilon_{n}^{(j)}} \left(g\left(\frac{i}{n}-s\right)-g\left(\frac{i-1}{n}-s\right)\right)W(\mathrm{d}s).$$

By assumption (CLT) and Lemma 2 we obtain the following inequalities (since σ is bounded on compact intervals)

$$\begin{split} \frac{1}{\sqrt{n}\tau_n^{p\wedge 1}} & \sum_{i=1}^{[nt]} \left(E|A_i^n|^2\right)^{\frac{p\wedge 1}{2}} \leq Cn^{-\gamma(p\wedge 1)+\frac{1}{2}} \\ \frac{1}{\sqrt{n}\tau_n^{p\wedge 1}} & \sum_{i=1}^{[nt]} \left(E|B_i^{n,\varepsilon_n^{(1)}}|^2\right)^{\frac{p\wedge 1}{2}} \leq Cn^{1/2}|\varepsilon_n^{(1)}|^{\gamma(p\wedge 1)} \\ \frac{1}{\sqrt{n}\tau_n^{p\wedge 1}} & \sum_{i=1}^{[nt]} \left(E|C_i^{n,\varepsilon_n^{(j)},\varepsilon_n^{(j+1)}}|^2\right)^{\frac{p\wedge 1}{2}} \leq Cn^{1/2}|\varepsilon_n^{(j+1)}|^{\gamma(p\wedge 1)}|\overline{\pi}^n(\varepsilon_n^{(j+1)}) - \overline{\pi}^n(\varepsilon_n^{(j)})|^{\frac{p\wedge 1}{2}}, \quad j = 1, \dots, l-1, \\ \frac{1}{\sqrt{n}\tau_n^{p\wedge 1}} & \sum_{i=1}^{[nt]} \left(E|C_i^{n,\varepsilon_n^{(i)},\varepsilon_n^{(l+1)}}|^2\right)^{\frac{p\wedge 1}{2}} \leq Cn^{1/2}\overline{\pi}^n(\varepsilon_n^{(l)})^{\frac{p\wedge 1}{2}}. \end{split}$$

Then we deduce by (CLT) and Lemma 3

$$\frac{1}{\sqrt{n}\tau_n^p}\sum_{i=1}^{[nt]} |\Delta_i^n Y - \sigma_{\frac{i-1}{n}} \Delta_i^n G|^p \xrightarrow{P} 0.$$

This completes the proof of Theorem 5.

8 Conclusion and outlook

In this paper we have derived convergence in probability and normal asymptotic limit results for multipower variations of processes Y that, up to a drift-like term, has the form

$$Y_t = \int_{-\infty}^t g(t-s) \,\sigma_s W(\mathrm{d}s)$$

where the kernel g is deterministic, $\sigma > 0$ is an adapted càdlàg process and W is the stochastic Wiener measure. A key type of example has g(t) behaving as t^{δ} for $t \downarrow 0$ and $\delta \in (-\frac{1}{2}, \frac{1}{2}) \setminus \{0\}$. In those instances Y is not a semimartingale and the limit theory of multipower variation developed for semimartingales does not suffice to derive the desired kind of limit results. The basic tool we establish and apply for this is a normal central limit theorem for triangular arrays of dependent Gaussian variables. As a case of some special interest for applications, particularly in turbulence, the central limit behaviour of the realised variation ratio, i.e. the ratio of bipower variation to quadratic variation, is briefly discussed. Some specific examples of choice of g are also considered.

The turbulence context referred to concerns time-wise observations of velocities at a single location x in space. More generally it would be of interest to develop the theory of multipower variation corresponding to a setting where velocities are observed along a curve τ in space-time. More specifically, suppose that velocity $Y_t(x)$ at position x and time t is defined by

$$Y_t(x) = \int_{A+(x,t)} g(t-s, x-\xi) \sigma_s(\xi) W(\mathrm{d}\xi \mathrm{d}s)$$

where W denotes white noise, $\sigma_t(x)$ is a positive stationary random field on \mathbb{R}^2 , g is a deterministic damping function and A is a subset of space-time involving only points with negative time coordinate. (For realism in turbulence modelling a drift term should be added to this expression for Y, but we ignore that here.) Then, with the curve τ parametrised as $\tau(w) = (x(w), t(w))$, say, the problem is to study multipower variations of the process X defined as

$$X_{w} = \int_{A+\tau(w)} g(t(w) - s, x(w) - \xi) \sigma_{s}(\xi) W(\mathrm{d}\xi \mathrm{d}s)$$

Among the questions that this raises is that of proper definition of filtrations. As to the latter, the concept of alignment, introduced in [BNSch09], is relevant. The definition is as follows.

Definition The curve τ and the ambit set A, with rectifiable and parametrised boundary $C = \{c(\gamma) : \gamma \in \Gamma\}$, are said to be aligned if the following conditions are satisfied. Let c^{\perp} denote the transversal of \dot{c} , i.e. $c^{\perp} = (\dot{c}_2, -\dot{c}_1)$.

- (i) For all w there exists a partition of C into two sets C_w^+ and C_w^- such that $\dot{\tau}(w) \cdot c^{\perp}(\gamma) \ge 0$ for all γ with $c(\gamma) \in C_w^+$ while $\dot{\tau}(w) \cdot c^{\perp}(\gamma) \le 0$ for all γ with $c(\gamma) \in C_w^-$.
- (ii) The subsets Γ_w^+ and Γ_w^- of Γ corresponding to C_w^+ and C_w^- are connected.
- (iii) For all w the curve lengths of C_w^+ and C_w^- are positive.

Note that the sets C_w^+ and C_w^- constitute the 'front' and the 'rear' of $A_{t(w)}(x(w))$ as (x(w), t(w)) moves along the curve τ .

In another direction it would be of interest to extend results of the present paper to power and multipower variations of higher order differences of Y. In particular, this might yield normal central limit theorems for the whole range of values of δ and it could also lead to more robustness against drift processes. For some recent work on quadratic variation of higher order differences, see [Beg07a] and [Beg07b] and references given there.

Appendix: Properties of the Bessel functions K

The Bessel functions K_{ν} are defined by

$$K_{\nu}(x) = \frac{1}{2} \int_{0}^{\infty} y^{\nu-1} e^{-\frac{1}{2}x(y^{-1}+y)} dy$$

where the index ν may take any real value. The known formulas for the K_{ν} recalled below are all given in [GrRy95].

A.1 Exact properties

Elementary exact properties are

$$K_{\nu}(x) = K_{-\nu}(x)$$
 (A.1.1)

$$K_{\nu+1}(x) = 2\nu x^{-1} K_{\nu}(x) + K_{\nu-1}(x)$$
(A.1.2)

$$K'_{\nu}(x) = -K_{\nu-1}(x) - \nu x^{-1} K_{\nu}(x)$$
(A.1.3)

Let

$$\bar{K}_{\nu}\left(x\right) = x^{\nu}K_{\nu}\left(x\right) \tag{A.1.4}$$

and, for $\nu > 0$,

$$\check{K}_{\nu}(x) = 2^{-\nu+1} \Gamma(\nu)^{-1} \bar{K}_{\nu}(x).$$
(A.1.5)

Here \bar{K}_{ν} may be reexpressed as

$$\bar{K}_{\nu}(x) = \frac{1}{2} \int_{0}^{\infty} y^{\nu-1} e^{-\frac{1}{2}y} e^{-\frac{1}{2}x^{2}y^{-1}} \mathrm{d}y, \qquad (A.1.6)$$

and, by a well known limit property of K_{ν} , we have (for $\nu > 0$)

$$\check{K}_{\nu}(x) \to 1 \quad \text{as} \quad x \downarrow 0.$$
 (A.1.7)

Note further that expressed in terms of \bar{K} , the relations (A.1.1), (A.1.2) and (A.1.3) take the form

$$\bar{K}_{\nu}(x) = x^{2\nu} \bar{K}_{-\nu}(x)$$
$$\bar{K}_{\nu+1}(x) = 2\nu \bar{K}_{\nu}(x) + x^2 \bar{K}_{\nu-1}(x)$$
(A.1.8)

and

$$\bar{K}'_{\nu}(x) = -x\bar{K}_{\nu-1}(x).$$
 (A.1.9)

The latter implies

$$\bar{K}_{\nu}''(x) = x^2 \bar{K}_{\nu-2}(x) - \bar{K}_{\nu-1}(x)$$

$$\bar{K}_{\nu}^{\prime\prime\prime}(x) = x\bar{K}_{\nu-2}(x) + 2x\bar{K}_{\nu-2}(x) - x^{3}\bar{K}_{\nu-3}(x)$$
$$= -(x^{3}\bar{K}_{\nu-3}(x) - 3x\bar{K}_{\nu-2}(x)).$$

We shall also need the following formula (formula 3.383.8 in [GrRy95]). For $t, \psi, \nu > 0$,

$$\int_{0}^{\infty} (t+s)^{\nu-1} s^{\nu-1} e^{-2\psi s} \mathrm{d}s = \sqrt{\frac{2}{\pi}} \Gamma\left(\nu\right) 2^{-\nu} \psi^{-\nu+\frac{1}{2}} t^{\nu-\frac{1}{2}} K_{\nu-\frac{1}{2}}\left(\psi t\right) e^{\psi t}.$$
 (A.1.10)

Finally, for $n = 0, 1, 2, \dots$ we have

$$K_{n+\frac{1}{2}}(x) = K_{\frac{1}{2}}(x) \left(1 + \sum_{i=1}^{n} \frac{(n+i)!}{i!(n-i)!} 2^{-i} x^{-i} \right)$$
(A.1.11)

with

$$K_{\frac{1}{2}}(x) = \sqrt{\frac{\pi}{2}} x^{-\frac{1}{2}} e^{-x}$$
(A.1.12)

while, when the index is a natural number,

$$K_{n}(x) = \frac{1}{2} \sum_{k=0}^{n-1} (-1)^{k} \frac{(n-k-1)!}{k!} \left(\frac{x}{2}\right)^{2k-n} + (-1)^{n+1} \sum_{k=0}^{\infty} \frac{1}{k! (n+k)!} \left[\log \frac{x}{2} - \frac{1}{2}\psi(k+1) - \frac{1}{2}\psi(n+k+1)\right] \left(\frac{x}{2}\right)^{2k+n} (8.1)$$

or, equivalently,

$$\bar{K}_{n}(x) = \frac{1}{2} \sum_{k=0}^{n-1} (-1)^{k} \frac{(n-k-1)!}{k!} \left(\frac{x}{2}\right)^{2k} + (-1)^{n+1} \sum_{k=0}^{\infty} \frac{1}{k! (n+k)!} \left[\log \frac{x}{2} - \frac{1}{2}\psi \left(k+1\right) - \frac{1}{2}\psi \left(n+k+1\right)\right] \left(\frac{x}{2}\right)^{2(k+n)} (8.2)$$

where ψ is the digamma function.

A.2 Asymptotic properties

The Bessel functions K_{ν} and I_{ν} are connected by

$$K_{\nu}(x) = \frac{1}{2} \frac{\pi}{\sin(\pi\nu)} \left(I_{-\nu}(x) - I_{\nu}(x) \right)$$
(A.2.1)

and we have

$$I_{\nu}(x) = \left(\frac{x}{2}\right)^{\nu} \sum_{n=0}^{\infty} \frac{\left(\frac{x}{2}\right)^{2n}}{n! \Gamma(n+\nu+1)}.$$

Using these relations and the fact that

$$\Gamma(1-\nu)\Gamma(\nu) = \frac{\pi}{\sin(\pi\nu)}$$
(A.2.2)

we find

$$\bar{K}_{\nu}(x) = 2^{\nu-1} \frac{\pi}{\sin(\pi\nu)} \left\{ \sum_{n=0}^{\infty} \frac{\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n-\nu+1)} - \left(\frac{x}{2}\right)^{2\nu} \sum_{n=0}^{\infty} \frac{\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n+\nu+1)} \right\}$$
$$= 2^{\nu-1} \frac{\pi}{\sin(\pi\nu)} \left\{ \frac{1}{\Gamma(1-\nu)} + \frac{1}{\Gamma(2-\nu)} \left(\frac{x}{2}\right)^{2} + \cdots - \frac{1}{\Gamma(\nu+1)} \left(\frac{x}{2}\right)^{2\nu} - \frac{1}{\Gamma(\nu+2)} \left(\frac{x}{2}\right)^{2(\nu+n)} - \cdots \right\}.$$

Alternatively we may write

$$\tilde{K}_{\nu}(x) = 1 + \sum_{n=1}^{\infty} \frac{\Gamma(1-\nu)\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n-\nu+1)} - \left(\frac{x}{2}\right)^{2\nu} \sum_{n=0}^{\infty} \frac{\Gamma(1-\nu)\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n+\nu+1)}.$$

$$= 1 + \sum_{n=1}^{\infty} \frac{\Gamma(1-\nu)\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n-\nu+1)} - \left(\frac{x}{2}\right)^{2\nu} \sum_{n=0}^{\infty} \frac{\Gamma(1-\nu)\left(\frac{x}{2}\right)^{2n}}{n!\Gamma(n+\nu+1)}.$$
(8.3)

From these expansions, distinguishing between the cases $0 < \nu < 1$ and $\nu > 1$, we obtain:

If $0 < \nu < 1$ then

$$\check{K}_{\nu}(x) = 1 - 2^{-2\nu} \frac{\Gamma(1-\nu)}{\Gamma(1+\nu)} x^{2\nu} + \frac{1}{4} \frac{1}{1-\nu} x^2 - O\left(x^{2(1+\nu)}\right).$$
(A.2.4)

When $\nu = 1$ we have, from (8.1),

$$\check{K}_{1}(x) = 1 - \frac{1}{2} \left| \log x \right| x^{2} + O(x^{2}).$$
(A.2.5)

For $\nu > 1$

$$\check{K}_{\nu}(x) = 1 - \frac{1}{4} \frac{1}{\nu - 1} x^2 - 2^{-2\nu} x^{2\nu} + O(x^4).$$
(A.2.6)

References

- [BNCP09] Barndorff-Nielsen, O.E., Corcuera, J.M. and Podolskij, M. (2009): Power variation for Gaussian processes with stationary increments, *Stoch. Proc. Appl.* 119, 1845-1865.
- [BNCPW09] Barndorff-Nielsen, O.E., Corcuera, J.M., Podolskij, M. and Woerner, J.H.C. (2009): Bipower variation for Gaussian processes with stationary increments, J. Appl. Prob. 46, 132-150.
- [BNGJPS06] Barndorff-Nielsen, O.E., Graversen, S.E., Jacod, J., Podolskij, M. and Shephard, N. (2006): A central limit theorem for realised power and bipower variations of continuous semimartingales. In: Yu. Kabanov, R. Liptser and J. Stoyanov (Eds.): *From Stochastic Calculus to Mathematical Finance. Festschrift in Honour of A.N. Shiryaev.* Heidelberg: Springer. Pp. 33-68.
- [BNGJS06] Barndorff-Nielsen, O.E., Graversen, S.E., Jacod, J. and Shephard, N. (2006): Limit theorems for bipower variation in financial econometrics, *Econometric The*ory 22, 677-719.
- [BNSch04] Barndorff-Nielsen, O.E. and Schmiegel, J. (2004): Lévy-based tempo-spatial modelling; with applications to turbulence. *Uspekhi Mat. NAUK* **59**, 65-91.

- [BNSch07] Barndorff-Nielsen, O.E. and Schmiegel, J. (2007): Ambit processes; with applications to turbulence and cancer growth. In F.E. Benth, Nunno, G.D., Linstrøm, T., Øksendal, B. and Zhang, T. (Eds.): Stochastic Analysis and Applications: The Abel Symposium 2005. Heidelberg: Springer. Pp. 93-124.
- [BNSch08a] Barndorff-Nielsen, O.E. and Schmiegel, J. (2008a): A stochastic differential equation framework for the timewise dynamics of turbulent velocities, *Theory Prob. Its Appl.* 52, 372-388.
- [BNSch08b] Barndorff-Nielsen, O.E. and Schmiegel, J. (2008b): Time change, volatility and turbulence. In A. Sarychev, A. Shiryaev, M. Guerra and M.d.R. Grossinho (Eds.): *Proceedings of the Workshop on Mathematical Control Theory and Finance*. Lisbon 2007. Berlin: Springer. Pp. 29-53.
- [BNSch08c] Barndorff-Nielsen, O.E. and Schmiegel, J. (2008c): Time change and universality in turbulence. Research Report 2007-8. Thiele Centre for Applied Mathematics in Natural Science. (Submitted.)
- [BNSch09] Barndorff-Nielsen, O.E. and Schmiegel, J. (2009): Brownian semistationary processes and volatility/intermittency. In Rungaldier, W., Schachermeyer, W. and **** (Eds.): ****. Pp. ***_***.
- [BNS02] Barndorff-Nielsen, O. E. and Shephard, N. (2002): Econometric Analysis of realised volatility and its use in estimating stochastic volatility models, J. Roy. Stat. Soc. B 64, 255-280.
- [BNS03] Barndorff-Nielsen, O. E. and Shephard, N. (2003): Realised power variation and stochastic volatility models. *Bernoulli* **9**, 243-265.
- [BNS04a] Barndorff-Nielsen, O. E. and Shephard, N.(2004a): Power and bipower variation with stochastic volatility and jumps (with discussion), J. Fin. Econometrics 2, 1-48.
- [BNS04b] Barndorff-Nielsen, O. E. and Shephard, N. (2004b): Econometric analysis of realised covariation: high frequency covariance, regression and correlation in financial economics, *Econometrica* 72, 885-925.
- [BNS06] Barndorff-Nielsen, O. E. and Shephard, N. (2006): Impact of jumps on returns and realised variances: econometric analysis of time-deformed Lévy processes. J. Econometrics 131, 217-252.
- [BNS07] Barndorff-Nielsen, O. E. and Shephard, N. (2007): Variation, jumps, market frictions and high frequency data in financial econometrics. In R. Blundell, T.

Persson and W.K. Newey (Eds.): Advances in Economics and Econometrics. Theory and Applications. Ninth World Congress. Econometric Society Monographs. Cambridge University Press. Pp. 328-372.

- [BNSW06] Barndorff-Nielsen, O. E. and Shephard, N. and Winkel, M. (2006): Limit theorems for multipower variation in the presence of jumps. Stoch. Proc. Appl. 116, 796-806.
- [Beg07a] Bégyn, A. (2007a): Asymptotic expansions and central limit theorem for quadratic variations of Gaussian processes. *Bernoulli* **13**, 712-753.
- [Beg07b] Bégyn, A. (2007b): Functional limit theorems for generalized quadratic variations of Gaussian processes. Stoch. Proc. Appl. 117, 1848-1869.
- [Bil68] Billingsley, P. (1968): Convergence of Probability Measures. New York: Wiley.
- [CNW06] Corcuera, J.M. Nualart, D. and Woerner, J.H.C. (2006): Power variation of some integral fractional processes. *Bernoulli* **12**, 713-735.
- [Cor07] Corcuera, J.M. (2007): Power variation analysis of some long-memory processes.
 In F.E. Benth, Nunno, G.D., Linstrøm, T., Øksendal, B. and Zhang, T. (Eds.): Stochastic Analysis and Applications: The Abel Symposium 2005. Heidelberg: Springer. Pp. 219-234.
- [CL67] Cramér, H. and Leadbetter, M.R. (1967): Stationary and Related Stochastic Processes. New York: Wiley.
- [Doo53] Doob, J.L. (1953): Stochastic Processes. Wiley, New York.
- [GrRy95] Gradshteyn, I.S. and Ryzhik, I.M. (1996): Table of Integrals, Series and and Products (Fifth Ed.) London: Academic Press.
- [Guy87] Guyon, X. (1987): Variations des champs gaussian stationaires: applications à l'identification. *Prob. Th. Rel. Fields* **75**, 179-193.
- [GuyLe89] Guyon, X. and Leon, J. (1989): Convergence en loi des H-variation d'un processus gaussien stationaire. Ann. Inst. Henri Poincaré **25**, 265-282.
- [HoSu87] Ho, H.C. and Sun, T.C (1987): A central limit theorem for non-instantaneous filters of a stationary Gaussian process, J. Mult. Anal. 22, 144-155.
- [HuNu05] Hu, Y. and Nualart, D. (2005): Renormalized self-intersection local time for fractional Brownian motion, Ann. Probab. **33**, 948-983.
- [Jac08a] Jacod, J. (2008a): Asymptotic properties of realized power variations and related functionals of semimartingales, *Stoch. Proc. Appl.* **118**, 517-559.

O.E. Barndorff-Nielsen, J.M. Corcuera and M. Podolskij: Asymptotic results

- [Jac08b] Jacod, J. (2008b): Statistics and high frequency data. Lecture notes.
- [Jac08c] Jacod, J. (2008c): Asymptotic properties of realized power variations and related functionals of semimartingales:: Multipower variations. Working paper.
- [JacTod09] Jacod, J. and Todorov, V. (2009): Testing for common arrival of jumps in discretely-observed multidimensional processes, *Ann. Stat.* (To appear.)
- [KiPo08] Kinnebrock, S. and Podolskij, M. (2008): A note on the central limit theorem for bipower variation of general functions. *Stoch. Proc. Appl.* **118**, 1056-1070.
- [Nab52] Nabeya, S. (1952): Absolute moments in 3-dimensional normal distribution. Ann. Inst. Stat. Math. 4, 15-30.
- [NoPec09] Nourdin, I. and Peccati, G. (2009): Non-central convergence of multiple integrals, Under revision for *Ann. Probab.*
- [Nu06] Nualart, D. (2006): *The Malliavin Calculus and Related Topics*. Berlin: Springer. (2nd edition.)
- [NuPec05] Nualart, D. and Peccati, G. 2005): Central limit theorems for sequences of multiple stochastic integrals, Ann. Probab. **33**, 177-193.
- [NuO-L08] Nualart, D. and Ortiz-Latorre, S. (2008): Central limit theorems for multiple stochastic integrals and Malliavin calculus, *Stoch. Proc. Appl.* **118**, 614-628.
- [PecTu05] Peccati, G. and Tudor, C.A. (2005): Gaussian limits for vector-valued multiple stochastic integrals. In: M. Emery, M. Ledoux and M. Yor (Eds), Seminaire de Probabilites XXXVIII. Lecture Notes in Mathematics 1857. Berlin: Springer. Pp. 247-262.
- [Taq77] Taqqu, M.S. (1977): Law of iterated logarithm for sums of non-linear functions of Gaussian variables that exhibit a long range dependence. Z. Wahrscheinlichkeitstheorie Verw. Geb. 40, 203-238.
- [Ver09] Veraart, A. (2009): Inference for the jump part of quadratic variation of Itô semimartingales. *Econometric Theory.* (To appear.)
- [Woe03] Woerner, J.H.C. (2003): Variational sums and power variation: a unifying approach to model selection and estimation in semimartingale models. *Statist. Decisions* **21**, 47-68.
- [Woe05] Woerner, J.H.C. (2005): Estimation of integrated volatility in stochastic volatility models. *Appl. Stochastic Models Business Industry.* **21**, 27-44.

- [Woe06] Woerner, J. H. C. (2006): Power and Multipower Variation: inference for high frequency data. A.N. Shiryaev, M. do Rosário Grossihno, P. Oliviera and M. Esquivel (Eds): Stochastic Finance. Heidelberg: Springer. Pp. 343-364.
- [Woe07] Woerner, J. H. C. (2007): Inference in Lévy type stochastic volatility models Adv. Appl. Prob. **39**, 531-549.
- [Woe08] Woerner, J. H. C. (2008): Volatility estimates for high frequency data: market microstructure noise versus fractional Brownian motion models. (Under revision)

Research Papers 2009



- 2009-7: Jeroen V.K. Rombouts and Lars Stentoft: Bayesian Option Pricing Using Mixed Normal Heteroskedasticity Models
- 2009-8: Torben B. Rasmussen: Jump Testing and the Speed of Market Adjustment
- 2009-9: Dennis Kristensen and Andrew Ang: Testing Conditional Factor Models
- 2009-10: José Fajardo and Ernesto Mordecki: Skewness Premium with Lévy Processes
- 2009-11: Lasse Bork: Estimating US Monetary Policy Shocks Using a Factor-Augmented Vector Autoregression: An EM Algorithm Approach
- 2009-12: Konstantinos Fokianos, Anders Rahbek and Dag Tjøstheim: Poisson Autoregression
- 2009-13: Peter Reinhard Hansen and Guillaume Horel: Quadratic Variation by Markov Chains
- 2009-14: Dennis Kristensen and Antonio Mele: Adding and Subtracting Black-Scholes: A New Approach to Approximating Derivative Prices in Continuous Time Models
- 2009-15: Charlotte Christiansen, Angelo Ranaldo and Paul Söderllind: The Time-Varying Systematic Risk of Carry Trade Strategies
- 2009-16: Ingmar Nolte and Valeri Voev: Least Squares Inference on Integrated Volatility and the Relationship between Efficient Prices and Noise
- 2009-17: Tom Engsted: Statistical vs. Economic Significance in Economics and Econometrics: Further comments on McCloskey & Ziliak
- 2009-18: Anders Bredahl Kock: Forecasting with Universal Approximators and a Learning Algorithm
- 2009-19: Søren Johansen and Anders Rygh Swensen: On a numerical and graphical technique for evaluating some models involving rational expectations
- 2009-20: Almut E. D. Veraart and Luitgard A. M. Veraart: Stochastic volatility and stochastic leverage
- 2009-21: Ole E. Barndorff-Nielsen, José Manuel Corcuera and Mark Podolskij: Multipower Variation for Brownian Semistationary Processes